



STATUTES & RULES

EMERGENCY PHONE +47 952 92 200

HEAD OFFICE, OSLO:

Assuranceforeningen SKULD (Gjensidig)

Ruseløkkvn. 26, 0251 Oslo, Norway

Mailing address:

P.O. Box 1376 Vika, 0114 Oslo, Norway

Telephone: +47 22 00 22 00
Telefax: +47 22 42 42 22
E-mail: osl@skuld.com

SKULD ABERDEEN:

Skuld Offshore Ltd.

1 Berry street, Aberdeen, AB25 1HF, UK
Telephone: +44 1224 84 37 15
E-mail: offshore@skuld.com

SKULD BERGEN:

Skuld Bergen

Nedre Ole Bulls plass 3, 5012 Bergen Mailing address: P.O. Box 75, 5803 Bergen, Norway

Telephone: +47 55 30 18 60
Telefax: +47 55 30 18 70
E-mail: bgo@skuld.com

SKULD COPENHAGEN:

Assuranceforeningen SKULD (Gjensidig)

ved Den Danske Afdeling Sankt Petri Passage 5, 3

DK-1165 Copenhagen K, Norway

Telephone: +45 33 43 34 00
Telefax: +45 33 11 33 41
E-mail: cph@skuld.com

SKULD HAMBURG:

Skuld Germay GmbH

Kurze Mühren 1

DE- 20095 Hamburg, Germany

Telephone: +49 40 309 9870
Telefax: +49 40 309 98716/98717
E-mail: germany@skuld.com

INFORMATION: S&R 02/2011

SKULD HONG KONG:

Skuld (Far East) Ltd.

Room 4107, China Resources Building 26 Harbour Road, Wanchai, Hong Kong

 Telephone:
 +852 2836 3909

 Telefax:
 +852 2836 3219

 E-mail:
 hkg@skuld.com

SKULD LONDON

Skuld UK

Forum House 15-18 Lime Street, 7th floor GB-London EC3M 7AN, UK

Telephone: +44 207 280 0110 E-mail: london@skuld.com

SKULD MOSCOW:

ZAO Skuld Insurance Eastern Europe (Skuld Russia)

Cernyshevskogo per.15 Moscow 127473, Russia Telephone: +7 495 657 96 48

Telefax: +7 495 657 96 49 E-mail: moscow@skuld.com

SKULD NEW YORK

Skuld North America Inc.

317 Madison Avenue, Suite 708

New York 10017, USA

Telephone: +1 212 758 9200 (24 hrs)
Telefax: +1 212 758 9935
E-mail: +1 212 758 9935
ny@skuld.com

SKULD PIRAEUS:

Skuld Hellas

Filellinon 1-3, 185 36 Piraeus P O Box 80112 GR-18510 Piraeus, Greece

Telephone: +30 210 459 4190-2 Telefax: +30 210 459 4194 E-mail: ath@skuld.com

www.skuld.com

CONTENTS

ST	TATUTES & RULES		
Th	ne Board of Directors		
Со	ommittee		
Management and Members of the Staff			
ST	TATUTES & RULES		
Sta	atutes	21	
1.	The association's purpose and organisation		
	Foundation and purpose	22	
	The General Meeting		
	Agenda of the Ordinary General Meeting		
	The Members' Committee		
	Functions of the Members' Committee	24	
	The Control Committee		
	Functions of the Control Committee		
	Board of Directors		
	Functions of the Board of Directors		
	The Association's elected representatives		
	The Election Committee		
2.	Disputes		
	Arbitration and Choice of Law	27	

3. Dissolution and period of transition

	36. Claims handling	69
	37. Time limits	
	38. Provision of security	
	39. Waiver and recovery	
	40. Exclusion of liability	
	42. Assignments and mortgages	
	43. Set-off	
	44. Interest	
	45. Joint members, co-assureds, affiliates and fleet entries	72
	46. Omnibus rule	
	47. Arbitration and law	74
A D	DENDIORS	
	PENDICES	
1.	Definitions	
2.	Terms of entry for co-assureds and affiliates	
3.	Overspill claims and calls	83
4.	Deductibles	89
5.	Oil pollution	91
5. A	A Passengers, crew and other persons carried on board	93
6.	Contracts and contractual terms	94
7.	Additional insurances	95
NO	RWEGIAN STATUTES / FORENINGENS VEDTEKTER	
1.	FORENINGENS FORMÅL OG ORGANISASJON	
	Stiftelse og formål	100
	Generalforsamlingen	
	Ordinær generalforsamlings gjøremål	
	Representantskapet	
	Representantskapets gjøremål	
	Kontrollkomitéen	
	Styret	
	Styrets gjøremål	
	Foreningens tillitsmenn	104
	Valgkomitéen	104
2.	TVISTER	
	Voldgift og lovvalg	104

STATUTES & RULES

3.	OPPLØSNING OG OVERGANGSREGLER		
	Oppløsning av Foreningen	105	
	Overgangsregler	105	
INI	DEX TO RULES		
	Index to Rules	108	

The Board of Directors 2011

Erik Gløersen, Chairman

Independent

Klaus Kjærulff, Vice Chairman

Smyril Line - KK Shipping

Manfredi Lefebvre D'Ovidio

Silversea Cruises Holding Ltd

Sergey Popravko

Unicom Management Services (Cyprus) Ltd (SCF Group)

Ulf G. Ryder

Stena Bulk AB

Trygve Seglem

Knutsen OAS Shipping AS

John P. Tavlarios

General Maritime Corporation

Peter Wilsund

A.P. Møller - Mærsk A/S

Nils Aardal

Jo Tankers AS

Simone A. Trondal

Skuld

Claus Spinding

Skuld



Committee 2011

Ulrich Niebusch, President

German Tanker Shipping GmbH & Co. KG

Yngvar Larsen, Vice President

Thor Dahl Management AS

Harald Andresen

Fred. Olsen & Co.

Capt. Cemil Bayülgen

U.N Ro-Ro Isletmeleri A.S

Bernt O. Bodal

American Seafoods Group

Knut Atle Ertsland

Continental Ship Management A/S

Tan Chin Hee

Pacific Carriers Ltd.

Frank G. Jensen

Clipper Group A/S

Huang Shao Jie

Associated Maritime Company (Hong Kong) Ltd.

Lou Kollakis

Chartworld Shipping Corporation

Dirk Lassen

Chemikalien Seetransport GmbH

Jacob Meldgaard

Torm A/S

Terje H. Michelsen

Grieg Shipping Group AS

Fabio Montanari

Navigazione Montanari SPA

Geraldine Otto

Seatrade Group

Marnix J. Van Overklift

Vomarine

Leon S Patitsas

Atlas Maritime Ltd.

Niels Stolberg

Beluga Shipping GmbH

Katy Therapiotis

Ionia Management S.A.

Felix H. Tschudi

Tschudi Shipping Company AS

Nicolas H. Vafias

Stealth Maritime SA

Hans Peter Westfal-Larsen

Westfal-Larsen Management A/S

Tian Zhongshan

Sinotrans Shipping Ltd.

Ann-Marie Åström

Rederi AB Gotland

MANAGEMENT AND STAFF MEMBERS' DIRECT AND/OR **MOBILE TELEPHONE NUMBERS**

To contact the Association after office hours kindly telephone the appropriate staff member. Some telephone numbers are listed below. For complete and always updated list of Skuld staff, please see Contact on www.skuld.com.

For emergencies, the Association maintains a telephone service outside office hours. Although this telephone service is at all times manned by a qualified claims executive, in all cases we recommend direct contact with the appropriate staff member.

EMERGENCY TELEPHONE NO. for all Skuld offices (international): +47 952 92 200

for Norway: 952 92 200

0SL0

MANAGEMENT

Jacobsohn, Douglas

Chief Executive Officer Direct: +47 22 00 22 01 Mobile: +47 952 92 201 E-mail: douglas@skuld.com

Hansen. Ståle

Chief Financial Officer Direct: +47 22 00 23 84 Mobile: +47 952 92 384 E-mail:

stale.hansen@skuld.com

Husebø, Trude S.

Executive Vice President Human Resources Direct: +47 22 00 22 06 Mobile: +47 952 92 206 F-mail:

trude.husebo@skuld.com

Jacobs, Eric

Executive Vice President General Counsel and Communication Direct: +47 22 00 22 71

Mobile: +47 952 92 271

E-mail:

eric.jacobs@skuld.com

Lindh, Captain Claes A.

Executive Vice President Underwriting & Marketing Direct: +47 22 00 22 28 Mobile: +47 952 92 228

F-mail.

claes.lindh@skuld.com

Westman, Claes

Executive Vice President Claims & Risk Management Direct: +47 22 00 22 25 Mobile: +47 952 92 225

F-mail:

claes.westman@skuld.com

Boger, Inger Helene

Executive Assistant for CEO Direct: +47 22 00 22 54 Mobile: +47 952 92 254 F-mail: ea@skuld.com

UNDERWRITING & MARKETING

Lindh, Captain Claes A.

Executive Vice President Underwriting & Marketing Direct: +47 22 00 22 28 Mobile: +47 952 92 228

E-mail:

claes.lindh@skuld.com

Bågenholm, Rune

Senior Vice President Underwriting & Marketing Direct: +47 22 00 22 10 Mobile: +47 952 92 210

F-Mail:

rune.bågenholm@skuld.com

Mandt. Egil Gulbrandsen

Senior Vice President Underwriting & Marketing Direct: +47 22 00 22 72 Mobile: +47 952 92 272

F-mail:

egil.mandt@skuld.com

Larsen. Ronny

Assistant Vice President Underwriting & Marketing Direct: +47 22 00 22 88 Mobile: +47 952 92 288

E-mail:

ronny.larsen@skuld.com

Bergh. Christina Polanscak

Senior Executive **Underwriting Support** Direct: +47 22 00 22 31 Mobile: +47 952 92 231

E-mail:

christina.bergh@skuld.com

LEGAL

Jacobs, Eric

Executive Vice President General Counsel and Communication Direct: +47 22 00 22 71 Mobile: +47 952 92 271

F-mail·

eric.jacobs@skuld.com

Hare, Jonathan

Senior Vice President Counsel Maritime Law & International Group Direct: +47 22 00 22 04 Mobile: +47 952 92 204

E-mail:

jonathan.hare@skuld.com

Løvskar, Hilde Søbstad

Maritime Trainee, Legal Direct: +47 22 00 22 69 Mobile: +47 952 92 269 E-mail: hilde.sobstad. lovskar@skuld.com

RISK MANAGEMENT

Risk Management

e-mail: ts@skuld.com

Lange, Paul-Bo

Senior Vice President Direct: +47 22 00 22 36 Mobile: +47 952 92 236

Bergh, Sverre

Senior Executive Direct: +47 22 00 22 38 Mobile: +47 952 92 238

Kleppe, Yngve

Executive Direct: +47 22 00 22 55 Mobile: +47 952 92 255

Langsrud. Hans Olav

Senior Executive Direct: +47 22 00 22 34 Mobile: +47 952 92 234

Blattner, Lene

Survey Co-ordinator Direct: +47 22 00 22 27 Mobile: +47 976 27 142

OFFSHORE

Offshore e-mail: offshore@skuld.com Offshore fax: +47 850 28 404

Thomas, Gregory

Senior Vice President Head of Offshore Direct: +47 22 00 22 48 Mobile: +47 952 92 248

Kobro. Christofer

Assistant Vice President Lawyer Client Servicing & Marketing Direct: +47 22 00 22 87 Mobile: +47 952 92 287

Zervens. Ann-Charlotte

Claims and Contracts Executive

Direct: +47 22 00 22 67 Mobile: +47 952 92 267

Kure-Dziakonski. Katinka

Client Servicing and Marketing Assistant Direct: +47 22 00 22 46 Mobile: +47 952 92 246

Ødegård, Anne Lise

Underwriting Assistant Direct: +47 22 00 22 58

OSLO SYNDICATE 1

Syndicate 1 e-mail: syn1.osl@skuld.com Syndicate 1 fax: +47 850 28 301

Dueled, Lars

Senior Vice President Head of Syndicate Direct: +47 22 00 23 12 Mobile: +47 952 92 312

Riley, Frank A.

Senior Vice President Defence

Direct: +47 22 00 22 65 Mobile: +47 952 92 265

Brevik, Hallgeir

Assistant Vice President Underwriting Direct: +47 22 00 22 18 Mobile: +47 952 92 218

Nerbø, Terje

Assistant Vice President Underwriting Direct: +47 22 00 22 17 Mobile: +47 952 92 217

Krüger, Jan-Erland

Assistant Vice President Underwriting Direct: +47 22 00 22 15 Mobile: +47 952 92 215

Mellilo. Flavia Pompa

Assistant Vice President Lawyer Claims Direct: +47 22 00 22 82 Mobile: +47 952 92 282

Trondal, Simone

Senior Claims Executive Lawyer Direct: +47 22 00 22 85 Mobile: +47 952 92 285

Deiligbierg, Wenche

Senior Claims Executive Direct: +47 22 00 23 13 Mobile: +47 952 92 313

Johansen, Unni

Senior Claims Executive Direct: +47 22 00 22 93 Mobile: +47 952 92 293

Lindfors, Marcus

Senior Claims Executive Direct: +47 22 00 22 49 Mobile: +47 952 92 249

Walldén, Sebastian

Senior Underwriting Executive

Direct: +47 22 00 22 16 Mobile: +47 952 92 216

Amundsen, Renée

Claims Executive Lawver

Direct: +47 22 00 22 90 Mobile: +47 952 92 290

Serracin, Ivan

Claims Executive Lawyer

Direct: +47 22 00 23 83 Mobile: +47 952 92 383

Sørlie, Inger Anne

Claims Executive Lawver

Direct: +47 22 00 22 80 Mobile: +47 952 92 280

Bjarnelöf-Sovtic, Stefan

Claims Executive Direct: +47 22 00 22 21 Mobile: +47 952 92 221

Falk, Christin

Assistant Underwriter Direct: +47 952 92 297 Mobile: +47 952 92 297

Flaten, Kari

Claims Executive Direct: +47 22 00 22 86 Mobile: +47 952 92 286

Ortiz de Rozas, Florencia

Claims Executive Direct: +47 22 00 23 42 Mobile: +47 952 92 342

Eivindsen, Mona

Assistant Underwriter Direct: +47 22 00 22 32 Mobile: +47 952 92 232

OSLO SYNDICATE 2

Syndicate 2 e-mail: syn2.osl@skuld.com Syndicate 2 fax: +47 850 28 302

Urdal, Ketil

Senior Vice President Head of Syndicate Direct: +47 22 00 22 75 Mobile: +47 952 92 275

Gutormsen, Tor Christian

Senior Vice President Underwriting Direct: +47 22 00 22 20

Direct: +47 22 00 22 20 Mobile: +47 952 92 220

Skuncke, Göran

Assistant Vice President Underwriting

Direct: +47 22 00 22 78 Mobile: +47 952 92 278

Hedqvist, Mattias

Assistant Vice President Lawyer Direct: +47 22 00 22 40

Mobile: +47 952 92 240

Thaulow, Yvonne

Underwriting Executive Direct: +47 22 00 22 24 Mobile: +47 952 92 224

Ivanov, Nikolai

Assistant Vice President Lawyer

Direct: +47 22 00 22 29 Mobile: +47 952 92 229

Kohmann, Kristoffer

Assistant Vice President, Claims Lawyer

Direct: +47 22 00 22 43 Mobile: +47 952 92 243

Horn, Kjeld B.

Senior Claims Executive Lawyer

Direct: +47 22 00 22 94 Mobile: +47 952 92 294

van Spriel, Inna

Senior Claims Executive Lawyer

Direct: +47 22 00 22 64 Mobile: +47 952 92 264

Carlmar, Marion

Senior Claims Executive Direct: +47 22 00 22 83 Mobile: +47 952 92 283

Rødsjø, Ragnhild

Senior Claims Executive Direct: +47 22 00 22 84 Mobile: +47 952 92 284

Jacobsen, Thomas André Hessel

Claims Executive Direct: +47 22 00 22 23 Mobile: +47 952 92 223

Nilsen, Cecilie Holm

Senior Claims Executive Direct: +47 22 00 22 26 Mobile: +47 952 92 226

Wikdal, Anne Ingeborg

Claims Executive Direct: +47 22 00 23 17 Mobile: +47 952 92 317

CREDIT CONTROL

Credit Control e-mail: Creditcontrol.osl@skuld.com

Bull, Elisabeth

Senior Executive Direct: +47 22 00 22 59 Mobile: +47 952 92 259 E-mail: elisabeth.bull@skuld.com

Mo, Marita

Credit Controller Direct: +47 22 00 23 49 Mobile: +47 952 92 349 E-mail: marita.mo@skuld.com

Nydahl, Merethe

Credit Controller Direct: +47 22 00 22 51

E-mail:

SKULD INFORMATION: S&R 02/2017

Opsahl, Gry Anita

Credit Controller Direct: +47 22 00 22 56 Mobile: +47 952 92 256

E-mail:

gry.anita.opsahl@skuld.com

Magnussen, Jan Elling

Credit Controller Direct: +47 22 00 22 70 Mobile: +47 952 92 270 E-mail: jan.elling. magnussen@skuld.com

Toterud, Karl Henrik

Credit Controller Direct: +47 22 00 22 79 Mobile: +47 952 92 279

F-mail:

karl.henrik.toterud@skuld.com

BERGEN

BERGEN

Syndicate e-mail: bgo@skuld.com Syndicate fax: +47 55 30 18 70

Flåm, Bjørn R.

Senior Vice President Head of Syndicate Direct: +47 55 30 18 65 Mobile: +47 952 92 365

Fjæreide, Berit E.

Senior Claims Executive Direct: +47 55 30 18 64 Mobile: +47 952 92 364

Frotvedt, Kjell Inge

Senior Claims Executive

Lawyer

Direct: +47 55 30 18 68 Mobile: +47 952 92 368

Valevatn, Kristian

Claims Executive Direct: +47 55 30 18 69 Mobile: +47 952 92 369

COPENHAGEN

MANAGEMENT

Lehmann, Helle

Senior Vice President Head of Office

Direct: +45 33 43 34 01 Mobile: +45 403 26 351

E-mail:

helle lehmann@skuld.com

Katkjær, Jan

Senior Vice President Deputy Head of Office Direct: +45 33 43 34 04 Mobile: +45 294 43 919 F-mail:

jan.katkjaer@skuld.com

Boje-Larsen, Michael

Vice President Lawver

Direct: +45 33 43 34 23 Mobile: +45 403 53 423 E-mail: michael.boielarsen@skuld.com

Vuylsteke, Sandro

Vice President Head of Claims

Direct: +45 33 43 34 14 Mobile: +45 401 06 861

F-mail:

sandro.vuylsteke@skuld.com

Gregersen, Stig

Vice President Head of FD&D

Direct: +45 33 43 34 10 Mobile: +45 402 50 961

F-mail:

stig.gregersen@skuld.com

Rasmussen, Kirsten

Management Assistant Direct: +45 33 43 34 03 E-mail: kirsten. rasmussen@skuld.com

Guldager, Vibeke

HR Executive Personal Assistant Direct: +45 33 43 34 50

E-mail:

vibeke.guldager@skuld.com

UNDERWRITING & MARKETING

Underwriting & Marketing e-mail: underwriting.cph@skuld.com

Katkjær, Jan

Senior Vice President Deputy Head of Office Direct: +45 33 43 34 04 Mobile: +45 294 43 919

Boie-Larsen. Michael

Vice President Lawver

Direct: +45 33 43 34 23 Mobile: +45 403 53 423

Lehmann. Helle

Senior Vice President

Head of Office

Direct: +45 33 43 34 01 Mobile: +45 403 26 351

Hansen, Lillian Frøhling

Underwriting Executive Direct: +45 33 43 34 16 Mobile: +45 206 36 473

Larsen, Birgitte

Underwriting Executive Direct: +45 33 43 34 25

CLAIMS AND FD&D

Claims & FD&D e-mail: claims.cph@skuld.com

Vuylsteke, Sandro

Vice President Head of Claims Direct: +45 33 43 34 14

Mobile: +45 401 06 861

Gregersen, Stig

Vice President Head of FD&D

Direct: +45 33 43 34 10 Mobile: +45 402 50 961

Rasch, Jørgen

Vice President Lawver

Direct: +45 33 43 34 20 Mobile: +45 404 06 522

Damgaard, Peter

Assistant Vice President Lawyer

Direct: +45 33 43 34 21 Mobile: +45 401 10 136 Home: +45 452 81 577

Zerman. Per

Assistant Vice President Lawver

Direct: +45 33 43 34 12 Mobile: +45 404 46 861

Larsen, Thomas K.

Senior Claims Executive Lawver

Direct: +45 33 43 34 42 Mobile: +45 213 41 855

Spinding, Claus

Senior Claims Executive Lawver

Direct: +45 33 43 34 40 Mobile: +45 213 41 868

Baert-Marquard, Valérie

Senior Claims Executive Direct: +45 33 43 34 13 Mobile: +45 204 61 982

Bengtsson, Sofia

Claims Executive Direct: +45 33 43 34 19

Mundt, Jonas Axelsen

Claims Executive Direct: +45 33 43 34 41 Mobile: +45 609 33 522

Naaman Jensen, Åse

Claims Executive Direct: +45 33 43 34 22 Mobile: +45 28 30 51 11

Petersen, Lena Holm

Claims Executive Direct: +45 33 43 34 17 Mobile: +45 305 69 589

Steffensen, Kim

Claims Executive Direct: +45 33 43 34 18 Mobile: +45 404 21 258

Wahl. Anders

Claims Executive Direct: +45 33 43 34 15 Mobile: +45 293 39 916

HAMBURG

HAMBURG

Syndicate e-mail: germany@skuld.com Phone: +49 40 309 9870 / Fax: +49 40 309 98716/17

Larsen. Thomas Biørn

Senior Vice President Head of Office Direct: +49 40 309 98721 Mobile: +49 151 1715 7017 E-mail: thomas.bjorn. larsen@skuld.com

Larsen, Ronny

Assisstant Vice President Mobile Norway: +47 952 92 288

Office: +49 40 309 98711 Mobile: +49 151 1715 7011 E-mail: ronny.larson@skuld.com

Krämer, Ulrich

Assistant Vice President Claims Direct: +49 40 309 98715 Mobile: +49 151 1715 7013 F-mail:

ulrich.kramer@skuld.com

Priess, Jens Michael

Assistant Vice President Lawver

Direct: +49 40 309 98723 Mobile: +49 151 1715 7019

F-mail:

jens.michael.priess@skuld.com

Clowes. Fiona

Senior Claims Executive Direct: +49 40 309 98724 Mobile: +49 151 1715 8031

F-mail.

Fiona.clowes@skuld.com

Hunstig, Alexandra

Claims Executive Direct: +49 40 309 98722 Mobile: +49 151 1715 7018 E-mail: Alexandra. hunstia@skuld.com

Eriksson, Johan Ivan

Claims Executive Direct: +49 40 309 98726 Mobile: +49 160 9028 9688 E-mail: johan.ivan. eriksson@skuld.com

Schaar, Dorothee

Senior Claims Executive Direct: +49 40 309 98714 Mobile: +49 151 1715 7014 F-mail:

Dorothee.schaar@skuld.com

Sonnenschein, Niklas Tjorben

Claims Executive Direct: +49 40 309 98727 Mobile: +49 170 227 3121 E-mail: niklas.

sonnenschein@skuld.com

Vitzthum, Simone

Claims Executive Direct: +49 40 309 98719 Mobile: +49 151 1715 7016

E-mail:

simone.vitzthum@skuld.com

Weyh, Olena

Claims Executive Direct: +49 40 309 98718 Mobile: +49 151 1715 7015

E-mail:

olene.weyh@skuld.com

Zamudio Santibanez, Angie Evelyn

Office Manager Direct: +49 40 309 98720 Mobile: +49 151 1715 7020

E-mail:

angie.zamudio@skuld.com

HONG KONG

HONG KONG

syndicate e-mail: hkg@skuld.com

Wang, Patrick

Senior Vice President Head of Syndicate Direct: +852 2836 4711 Mobile: +852 9408 1858 E-mail: adm.hkg@skuld.com

Mason, Nicola

Vice President Deputy Head of Syndicate Direct: +852 2836 4715 Mobile: +852 9028 4217

Hall, Christopher

Vice President, Marketing Senior Lawyer Direct: +852 2836 4710 Mobile: +852 9883 8874

Ott, Christian

Assistant Vice President Direct: +852 2836 4723 Mobile: +852 9193 7415

Fan, Wei

Claims Executive Lawyer Direct: +852 2836 4713 Mobile: +852 9105 7036

Ho, Keng Hoong

Senior Claims Executive Lawver

Direct: +852 2836 4718 Mobile: +852 9022 3786

Lau, Rita

Senior Claims Executive Lawyer Direct: +852 2836 4721 Mobile: +852 9022 7625

Chu, Jenny

Claims Executive Direct: +852 2836 4716 Mobile: +852 9885 8380

Dubey, Binoy Kumar Captain

Claims Executive Direct: +852 2836 4722 Mobile: +852 9766 7810

Yeung, Jenny

Office Manager Direct: +852 2836 4712 Mobile: +852 9880 6034 E-mail: adm.hkg@skuld.com

LONDON

LONDON

e-mail: london@skuld.com

Gattberg, Stefan

Senior Executive Deputy Head of London operations Direct: +44 207 280 0112 Mobile: +44 770 2737 338

E-mail:

stefan.gattberg@skuld.com

MOSCOW

MOSCOW

Representative Office e-mail: moscow@skuld.com Office phone: +7 495 657 96 48/ Fax: +7 495 657 96 49

Kamanin, Sergey

Vice President Head of Office

Mobile: +7 495 796 64 85

E-mail:

sergey.kamanin@skuld.com

Atrash, Darya

Office Manager Mobile: +7 926 702 70 82 E-mail: darya.atrash@skuld.com

NEW YORK

NEW YORK

email: ny@skuld.com

Anderson, Charles

Senior Vice President Head of Skuld North America

Direct: +1 212 758 9936 Mobile: +1 914 391 7908

Alamina, Renan

Vice President Direct: +1 212 758 9938

Mobile: +1 516 639 1585

Milgrim, Pamela

Vice President Direct: +1 212 758 9937 Mobile: +1 917 836 8181

Leikvang, Ulrika

Assistant Vice President Lawyer Direct: +1 212 935 8061 Mobile: +1 646 409 9422

Nelson, Mindy

Claims Assistant Direct: +1 212 935 8330 Mobile: +1 646 319 5076

PIRAEUS

PIRAEUS

syndicate e-mail: ath@skuld.com

Rydén, Clas

Senior Vice President Head of Office Direct: +30 210 459 4181 Mobile: +30 6944 605 539

Souli, Elina

Assistant Vice President Lawyer Direct: +30 210 459 4182 Mobile: +30 6934 500 351

Valentin, Charlotte

Assistant Vice President Lawyer Direct: +30 210 459 4185 Mobile: +30 6932 587 152

Behnke, Alexander

Claims Executive Direct: +30 210 459 4188 Mobile: +30 6937 178 881

Kotsakis, Leandros

Claims Executive Direct: +30 210 459 4186 Mobile: +30 6932 100 076

McDowall, Clare

Claims Executive Direct: +30 210 489 4187 Mobile: +30 6937 178 880

Paranos, Olivia

Claims Executive Direct: +30 210 459 4183 Mobile: +30 6937 100 518

Samaritis, Konstantinos

Claims Executive Direct: +30 210 459 4184 Mobile: +30 6932 100 075

Sirimi, Athina

Marketing Executive Direct: +30 210 459 4189 Mobile: +30 6932 100 074

For complete and always updated list of all Skuld staff members, please see www.skuld.com

STATUTES & RULES

PREFACE

These Statutes and Rules, which reflect the Association's membership in the International Group of P&I Associations and as a party to the International Group's Pooling Agreement, shall apply for the policy year 2011-2012 and come into effect from 20 February 2011.

Statutes



1. THE ASSOCIATION'S PURPOSE AND ORGANISATION

1.1 Foundation and purpose

- 1.1.1 Assuranceforeningen Skuld (Gjensidig) was founded in January 1897, and has its head office in Oslo, with offices in Bergen, Copenhagen, Hamburg, Hong Kong and Piraeus.
- 1.1.2 Membership in the Association is open to shipowners, operators, disponent owners, insurers and charterers of ships including such mobile floating structures as may from time to time be decided by the Board of Directors.
- 1.1.3 The purpose of the Association is mutual insurance against liabilities and losses incurred by members in direct connection with the operation of the entered vessels, including any business related to such insurance, hereunder defence insurance, insurance of charterers and insurance of mobile offshore units and similar mobile floating structures.
- 1.1.4 The Association may participate in reinsurance and coinsurance arrangements that fall within the purpose of the Association.
- 1.1.5 The Association may establish subsidiary companies to carry out business that falls within the purpose of the Association.
- 1.1.6 The Association may appoint a Manager.
- 1.1.7 Members shall not be liable to other members or to third parties for the liabilities of the Association.
- 1.1.8 When leaving the Association, members are not entitled to any share of the Association's equity or surplus.
- 1.1.9 The Association's Policy year and Accounting year shall terminate at 12.00 (noon) GMT on 20 February each year.
- 1.1.10 The members are, together with the Association, also members of Skuld Mutual Protection and Indemnity Association (Bermuda) Ltd. The membership is subject to the conditions of Bermuda Statutes 1978 No. 16 with later amendments.

1.2 The General Meeting

- 1.2.1 The General Meeting is the Association's highest authority.
- 1.2.2 The ordinary General Meeting shall be held within 6 months after the expiry of the accounting year. An extraordinary General Meeting shall be held whenever the Board of Directors or the Members' Committee considers it necessary or when at least 10 members who hold between them a minimum of 20 votes so demand.
- 1.2.3 Venues for the General Meeting are in addition to Oslo: Copenhagen, Piraeus, Hong Kong, Singapore, London and Hamilton, Bermuda.
- 1.2.4 The General Meeting is called by the Board of Directors upon at least 14 days written notice. The notice of the meeting shall include the Members' Committee's and Board of Directors' reports on the items which are on the agenda and such documents as the Board of Directors find necessary. The notice of the ordinary General Meeting shall include the Board of Directors' proposal for the annual accounts accompanied by the statement of the Auditors', the Members' Committee's and the Control Committee's report.
- 1.2.5 The General Meeting may only decide on matters set out in the notice calling the meeting and on which the Members' Committee has had the opportunity to put forward a report.

- 1.2.6 The General Meeting shall be opened by the Chairman of the Board, or in his absence by a person appointed by the Board of Directors. The General Meeting appoints its chairman. The General Meeting shall form a quorum regardless of the number of members present.
- 1.2.7 The entered tonnage of each member shall entitle the member to a maximum number of votes calculated on the total entered gross tonnage:

1 up to and including	20,000 one vote	
up to and including	50,000 a total of 2 votes	
up to and including	100,000 a total of 3 votes	
up to and including	200,000 a total of 4 votes	
up to and including	400,000 a total of 5 votes	
up to and including	600,000 a total of 6 votes	
up to and including	800,000 a total of 7 votes	
up to and including	1,000,000 a total of 8 votes	

Each 500,000 GT in excess of the first 1,000,000 GT shall entitle the member to 1 additional vote.

- 1.2.8 Entries for a period of less than one year or entries not subject to the applicable accounting year's supplementary call shall not give the right to vote. The right to vote shall be calculated on the recorded tonnage entered on the day on which the notice of meeting is sent to members. Voting rights shall not, however, be calculated on vessels which have in the meantime ceased to be entered.
- 1.2.9 Members whose vessels are managed or insured by one firm of operators or insurers hold between them as many votes as they would have held if all the entered vessels had belonged to one member.
- 1.2.10 Every member may be represented by a proxy at the General Meeting. Only members with voting rights according to 1.2.8 are eligible to act as proxy. No one may, however, cast votes which amount to more than 1/10 of the total votes present at the meeting.
- 1.2.11 The call to the General Meeting may require the participants of the Meeting to give notice in advance as to which member they will represent. Such notice shall be made in writing and must have arrived during office hours at least two days before the General Meeting.
- 1.2.12 Decisions are arrived at by a simple majority vote. However, to amend these Statutes a two-thirds majority is required. In the event of an equal number of votes being cast the chairman of the meeting shall have the deciding vote.

1.3 Agenda of the Ordinary General Meeting

The ordinary General Meeting shall:

- 1.3.1 decide on the adoption of the Statement of Accounts and the Board of Directors' Report, and decide on the level of supplementary calls, reimbursement of excess advance calls or the distribution of any surplus. The General Meeting may not stipulate lower supplementary calls, reimbursement of higher levels of excess advance calls or distribution of a larger part of the surplus than that proposed by the Members' Committee.
- 1.3.2 consider the Report from the Control Committee,
- 1.3.3 determine the number of members to serve on the Members' Committee,
- 1.3.4 elect members of the Members' Committee and determine their remuneration.

- 1.3.5 elect members of the Election Committee,
- 1.3.6 elect members of the Control Committee and determine their remuneration,
- 1.3.7 decide on the closing of policy years,
- 1.3.8 decide on amendments to the Statutes of the Association, and
- 1.3.9 decide on other matters within the authority of the General Meeting and stated in the agenda.

1.4 The Members' Committee

- 1.4.1 The Members' Committee shall comprise, in addition to a President and a Vice President, a minimum of 10 and a maximum of 28 members in a total number as determined by the General Meeting. The members of the Members' Committee are elected by the General Meeting after having considered the proposals of the Election Committee. The President and Vice President of the Members' Committee shall be elected separately.
- 1.4.2 The Members' Committee shall form a quorum when at least half of the members, including the President or the Vice President, are present.
- 1.4.3 Members' Committee meetings are held as often as the President deems necessary or at the request of at least 1/3 of the Members' Committee Members, or at the request of the Board of Directors.
- 1.4.4 The members of the Board of Directors and the Association's Managing Director (CEO) shall participate at meetings of the Members' Committee unless it decides otherwise for a particular matter. They shall participate without voting rights.

1.5 Functions of the Members' Committee

- 1.5.1 The Committee shall supervise the conduct of the business of the Board of Directors and the Managing Director (CEO) and ensure that the Association's purpose is promoted in accordance with applicable legislation, the Statutes and Resolutions by the General Meeting and the Members' Committee. Each of the Committee members shall have the right at the Meeting to request information about the Association's business. The Members' Committee may itself or by appointed representatives initiate investigations.
- 1.5.2 The Members' Committee shall, upon recommendations from the Election Committee, elect the members of the Board of Directors, including by separate election its Chairman and Vice Chairman and determine the Board of Directors' remuneration.
 - The Members' Committee shall also, upon recommendations from the Election Committee, elect the members of the Board of Directors of all subsidiaries or associated companies of the Association.
- 1.5.3 The Members' Committee shall also:
 - a) submit a report to the General Meeting as to whether the Board of Directors' proposals for the Statement of Accounts, the Board of Directors' Report and the application of loss or profit, hereunder the level of supplementary calls, reimbursement of excess advance calls or the distribution of any surplus, shall be adopted,
 - submit to the General Meeting its recommendations regarding the closing of policy years,

- c) submit a report to the General Meeting concerning amendments to the Statutes recommended by the Board of Directors,
- d) determine the Instructions to and examine the Report from the Control Committee.
- e) elect an Auditor and approve his remuneration,
- f) approve management contracts,
- g) deal with any other matters within the authority of the Members' Committee and stated in the agenda.

1.6 The Control Committee

- 1.6.1 The Control Committee shall consist of at least three members and one alternative member elected by the General Meeting.
- 1.6.2 The Control Committee shall elect each year its own Chairman from among its members.
- 1.6.3 The Control Committee shall form a quorum when at least three of its members are present and reasonable prior notice of the meeting has been given to its members.

1.7 Functions of the Control Committee

The Control Committee shall in accordance with the instructions given by the Members' Committee supervise the activities of the Association and ensure that it complies with the applicable laws, regulations and conditions, as well as with the resolutions and rules of the Association's decision-making bodies. The Control Committee shall submit a Report at least once a year to the Members' Committee and the General Meeting.

1.8 Board of Directors

- 1.8.1 The Board of Directors shall see to it that the Association's and its members' interests are being attended to in a proper way, and make such decisions as are not within the province of the Members' Committee or the General Meeting.
- 1.8.2 The Board of Directors of the Association shall, in accordance with a decision made by the Members' Committee, consist of 5 to 9 members including a Chairman and Vice Chairman. They shall be elected without substitutes in accordance with Statute 1.5.2. In addition, two members being elected by and among the employees of the Association or of a subsidiary appointed as Manager shall sit on the Board
- 1.8.3 The Board shall convene by notice from the Chairman. The Chairman may decide that matters shall be submitted in writing or dealt with in any satisfactory way. A Board member and the Managing Director (CEO) may demand that the Board of Directors shall deal with a particular matter.
- 1.8.4 The Board of Directors forms a quorum when more than half of its members are present or participate in the handling of a matter. In order to have a valid decision, those who support a decision must comprise more than 1/3 of the whole Board. In the event of an equal number of votes being cast, the Chairman or in his absence the Vice Chairman shall have the deciding vote.
- 1.8.5 The Managing Director (CEO) of the Association shall participate in the meetings of the Board of Directors, unless the Board decides otherwise for a particular matter. He shall participate without voting rights.

1.9 Functions of the Board of Directors

The Board of Directors shall:

- 1.9.1 administer the daily business of the Association.
- 1.9.2 appoint the Association's Managing Director (CEO) and determine his authority and conditions of service.
- 1.9.3 make decisions concerning branch offices and subsidiaries and determine the general operating principles for their activity. Where a branch office is established under a local board of directors then directives shall be issued in consultation with and on the recommendation of such local board. The authority of the local board shall conform with the principles outlined by the Association's Board of Directors,
- 1.9.4 submit to the Members' Committee a complete and audited Statement of Accounts and the Board of Directors' Report, hereunder recommendations regarding closing of policy years, the level of supplementary calls, reimbursement of excess advance calls or the distribution of any surplus,
- 1.9.5 submit to the Members' Committee recommendations regarding changes to the Statutes.
- 1.9.6 decide on or amend the Association's insurance conditions ("Rules").
- 1.9.7 approve the Association's reinsurance contracts,
- 1.9.8 administer the funds of the Association in accordance with adopted principles,
- 1.9.9 decide on release calls.
- 1.9.10 decide on the general premiums for the insurance year,
- 1.9.11 establish general principles for the approval of compensation payments,
- 1.9.12 in a particular case decide whether the Association shall compensate liabilities or losses beyond those provided for under the Association's insurance conditions ("Rules") or excluded from cover therein when the Board of Directors considers in accordance with the purpose of the Association that such compensation would be natural and desirable.
- 1.9.13 make recommendations to the Members' Committee and the General Meeting concerning any matter submitted for consideration.

1.10 The Association's elected representatives

- 1.10.1 Any person who is the appointed representative of a member shall be eligible for election to the Members' Committee and the Board of Directors. In addition one person who is not otherwise eligible may be elected as a member of the Board of Directors.
- 1.10.2 Any appointed representative of a member of the Association, except the President and Vice President of the Members' Committee and the members of the Board, is eligible for election to the Control Committee. In addition, one person who is not otherwise eligible may be elected. One member of the Control Committee shall have the qualifications required by Norwegian Law for appointment as Judge in the Law Courts. The election of this member shall be subject to approval by the Financial Supervisory Authority of Norway ("Kredittilsynet").
- 1.10.3 Members of the Members' Committee, the Control Committee and Board of Directors shall normally serve for two years. Approximately one half of the members shall retire from office each year. Re-election of members is permitted.
- 1.10.4 Members of the Board of Directors and the Members' Committee who no longer qualify as stipulated in 1.10.1 shall no longer hold office.

1.10.5 No one can hold office at the same time on the Board of Directors and the Members' Committee. No one can be elected or reelected after having attained the age of 67 years. The individual may, however, continue to serve the remainder of the period for which he was elected.

1.11 The Election Committee

- 1.11.1 The Association's Election Committee shall comprise of the President of the Members' Committee, and two appointed representatives of the members elected by the General Meeting and the Managing Director (CEO) serving as Secretary with the right to participate and speak. The Election Committee elects its Chairman.
- 1.11.2 The Election Committee shall make recommendations for the election of all the Association's elected representatives.
- 1.11.3 In its recommendation the Election Committee shall take into account that the composition of the Board of Directors and the Members' Committee shall reflect the Association's international activities, tonnage entered and structure of membership, and be comprehensively composed.
- 1.11.4 The Board of Directors shall approve Guidelines for the Election Committee.

2. DISPUTES

2.1 Arbitration and Choice of Law

- 2.1.1 These Statutes (Bye-Laws) are governed by Norwegian Law. The Insurance Contracts Act of 1989 shall not apply.
- 2.1.2 If nothing is agreed to the contrary, disputes between the Association and a member or a former member or a co-assured or a non-member who is or has been held covered under a policy issued by the Association in connection with his membership or insurance shall be decided by Arbitration. Each party shall appoint an arbitrator who shall together appoint a chairman of the arbitration tribunal. If the two arbitrators are unable to agree on the appointment of a chairman or one party refrains from exercising an appointment then he shall be appointed by the Chief Justice of the Oslo City Court.
- 2.1.3 The arbitration proceedings shall take place in Oslo and Norwegian law shall prevail. The arbitration decisions shall be reasoned.

3. DISSOLUTION AND PERIOD OF TRANSITION

3.1 Dissolution of the Association

The dissolution of the Association or amalgamation with other associations shall be determined at a General Meeting by a majority of at least two-thirds of the total registered membership calculated in accordance with 1.2.7 and the following. If less than two-thirds of the votes are present the matter shall be referred to a subsequent General Meeting and the proposal to dissolve or amalgamate the Association shall be decided by a majority of not less than three-quarters of the votes at the meeting. On dissolution of the Association, any surplus realised after discharge of the obligations of the Association, shall be disposed of as decided by the General Meeting. In the event of amalgamation any such surplus may at the



discretion of the General Meeting be disposed of as mentioned above or transferred to the new association.

3.2 Provisions for the period of transition

These Statutes shall come into force and the previous Statutes shall cease to have effect as from 20 February 2011 12.00 (noon) GMT. Liabilities or losses incurred before such time shall be determined in accordance with the Statutes then applicable.

N.B. The Statutes in the Norwegian language appear after the Appendices.

The Norwegian text shall take precedence.

Rules

Part I

MEMBERSHIP, COVER AND PREMIUMS



1. MEMBERSHIP

1.1 Application and entry

- 1.1.1 An owner or a charterer of a vessel may apply for entry of the vessel in the Association.
- 1.1.2 The Association may, in its absolute discretion, accept the application subject to specified conditions or restrictions, or reject the application, without providing any reasons.
- 1.1.3 If the entry is accepted, the owner or charterer shall become a member in the Association, the vessel shall become an entered vessel and the Association shall issue a Certificate of Entry.
- 1.1.4 Unless otherwise agreed, the entry of the vessel and the insurance cover shall commence at 12.00 hours GMT on the date the entry is accepted and continue until 12.00 hours GMT on the next 20 February, and thereafter from policy year to policy year, unless and until the cover ceases or the entry is terminated in accordance with these Rules

1.2 Joint members, co-assureds, affiliates and fleet entries

- 1.2.1 The Association may accept an entry of a vessel on behalf of more than one owner in which case each party shall be a joint member of the Association.
- 1.2.2 The Association may agree to extend the insurance cover provided to the member to a co-assured named in the Certificate of Entry or to an affiliate who shall not be named in the Certificate of Entry.
- 1.2.3 Co-assureds and affiliates shall not be members of the Association and their terms of entry shall be subject to the requirements of the International Group of P&I Association's Pooling Agreement, as set out in Appendix 2.
- 1.2.4 The Association may accept the entry of more than one vessel as a fleet entry.
- 1.2.5 With respect to liability for premiums and other sums due, see Rule 45.

1.3 Duration of membership

The member shall remain a member of the Association until the cover ceases or the entry is terminated, in respect of all vessels which the member has entered.

1.4 Definitions

The words "owner", "charterer", "vessel", "co-assured", "fleet entry" and "affiliate" shall have the meanings set out in Appendix 1, which also contains the definitions of other words used in the Rules.

2. TERMS OF COVER

2.1 Conditions of cover

The terms of entry and cover provided by the Association to the member are subject to the Statutes, these Rules and any special conditions agreed between the member and the Association.

Provided always that it is not intended that any third party other than those referred to in Rules 1.2 or permitted assign has any right under this Contract.

2.2 Risks covered

- 2.2.1 The member is only covered for such of those risks specified in Parts II (P&I cover), III (Defence cover) and Appendix 7 (Additional Insurances) of these Rules as are expressly agreed between the member and the Association.
- 2.2.2 The cover provided in Parts II, III and Appendix 7 is subject to the limitations and other terms set out in Parts I and IV and Appendices 1 6.

2.3 Scope of cover

The member is only covered in respect of liabilities, losses, expenses and costs incurred by the member which arise.

- 2.3.1 in direct connection with the operation of the entered vessel by the member in the member's capacity as owner or charterer of the vessel,
- 2.3.2 in respect of the member's interest in the entered vessel, and
- 2.3.3 out of events occurring during the period of entry of the vessel for the relevant risk in the Association.

3. CESSATION OF COVER AND TERMINATION

3.1 Cessation of cover

Unless the Association agrees otherwise, the insurance cover shall cease immediately where,

- 3.1.1 there is a change of management or ownership of the entered vessel,
- 3.1.2 the member, being an individual, becomes bankrupt, has a receiving order made against him, or becomes insolvent,
- 3.1.3 the member, being a corporation, is dissolved, wound up, has a receiver or liquidator appointed or commences proceedings under any bankruptcy or insolvency laws to seek protection from its creditors,
- 3.1.4 the entered vessel becomes a total loss, or is accepted by the hull underwriters or deemed by the Association as being a constructive, compromised or arranged total loss, except in respect of liability arising out of the casualty which gives rise to the total loss.
- 3.1.5 the vessel is missing for ten days from the date she was last heard of,
- 3.1.6 the vessel is posted at Lloyd's as missing, or
- 3.1.7 the vessel is requisitioned by a State or Government Authority, except that the insurance cover shall be reinstated after the period of requisition has ceased.

3.2 Termination by the member

The member may terminate the entry of any vessel with effect from 12.00 hours GMT on 20 February by giving written notice of termination prior to 20 January.

3.3 Termination by the Association

- 3.3.1 The Association may terminate the entry of any vessel with effect from 12.00 hours GMT on 20 February by giving written notice of termination prior to 20 January.
- 3.3.2 The Association may also terminate the entry of any or all vessels entered by the member or on behalf of more than one member.



- a) on immediate notice, where the member is in breach of his obligations under Rule 28.1 (in respect of disclosure and alteration of risk), Rule 28.2 (in respect of fraudulent, unlawful or deliberate acts) or Rule 28.4 (in respect of the classification and certification of the vessel).
- b) on three days' notice, where the member is in breach of his obligations under Rule 28.3 (in respect of the payment of premiums and other sums due to the Association).
- c) on seven days' notice, where the vessel is unseaworthy and the member has not made her seaworthy without undue delay, or where the member has not allowed the Association to carry out a survey in accordance with Rule 35, or where the member has notified the Association of any change of circumstance which materially alters the risks covered by the Association, or
- d) on thirty days' notice, without giving any reason.

3.4 Effect of Cessation and Termination

- 3.4.1 The Association shall be under no liability whatsoever in respect of any event occurring after cessation or termination.
- 3.4.2 Where cover ceases under Rule 3.1 (cessation of cover) or where the entry is terminated under Rule 3.3.2(c) or (d) (termination by the Association on seven or thirty days' notice), the member shall be entitled to a pro rata reduction of premiums in respect of the then policy year (for the period from cessation or termination until the end of the policy year), except that no reduction shall be allowed in respect of overspill calls.
- 3.4.3 Where cover ceases or the entry is terminated or cover is otherwise restricted or lost under these Rules, the member shall (subject to the reduction provided in Rule 3.4.2) remain liable for all premiums in respect of the then current policy year and all prior policy years.
- 3.4.4 Where cover ceases or the entry is terminated, the Association may nevertheless, in its absolute discretion, reinstate the entry and cover with effect from such time as it considers appropriate.

4. PREMIUMS, RESERVES AND DEDUCTIBLES

4.1 Premiums Generally

- 4.1.1 The premiums for each entered vessel shall be determined by the Association, taking into account all matters which the Association considers relevant, including the member's loss record.
- 4.1.2 The premiums determined by the Association and payable by the member may include the following,
 - a) annual calls
 - b) supplementary calls
 - c) overspill calls
 - d) release calls
 - e) fixed premiums
 - f) additional insurance premiums.

4.2 Estimated Total Calls (ETC)

- 4.2.1 The Association shall in each policy year determine a full ETC for each entered vessel, which shall constitute the full annual payable premium for that vessel, subject to any supplementary, overspill or release calls which may from time to time be determined by the Association as set out below
- 4.2.2 The ETC for vessels entered after the commencement of the policy year shall be calculated pro rata.

4.3 Supplementary Calls

- 4.3.1 The Association may for each open policy year determine supplementary calls to cover the liabilities of the Association (excluding any liability for overspill calls as defined in Appendix 3), the Association's running costs and an appropriate allocation to reserves in respect of open and closed policy years.
- 4.3.2 Any supplementary calls shall be levied on a pro rata basis on the ETC for the policy year.
- 4.3.3 The Association may levy additional supplementary calls on open policy years if considered necessary.
- 4.3.4 Separate supplementary calls may be levied for Defence cover.

4.4 Overspill Calls

The Association may levy one or more overspill calls in respect of an overspill claim in accordance with Appendix 3.

4.5 Release Calls

- 4.5.1 Where cover ceases or the entry is terminated for any one vessel, the Association may determine a release call for open policy years based on the anticipated calls for those years and such other factors as the Association considers relevant.
- 4.5.2 If the member pays the release call in full, the member shall not be obliged to pay any further premiums to the Association except for overspill calls, but the member shall not be entitled to receive payment of any surplus or reimbursement of any excess premium.
- 4.5.3 The member may, instead of paying the release call, irrevocably elect to pay the supplementary calls as subsequently adopted by the General Meeting every year prior to the closing of the policy year, in which event the member must within four weeks after the Notice of Release Call deposit with the Association an amount which the Association in its absolute discretion considers to be adequate to cover the anticipated supplementary calls for all open years. After the lapse of said four weeks, the release call shall be paid in full.

4.6 Fixed Premiums

- 4.6.1 The Association may agree fixed premiums for charterers and special categories of members.
- 4.6.2 Where fixed premiums have been agreed, the member shall not pay annual calls, supplementary calls, overspill calls or release calls, and the member shall not be entitled to any surplus.



4.7 Lay-up Returns

- 4.7.1 When the entered vessel is laid up in a safe port or place approved by the Association without cargo and without crew (other than for maintenance and security) on board for 30 consecutive days or more, the member shall be entitled (subject to any special terms which may have been agreed) to a pro rata reduction or return of premiums (other than overspill calls) in respect of the laid up period, in such proportion as the Association considers appropriate, provided that written notice of the lay-up is given to the Association no later than three months after the end of lay-up.
- 4.7.2 If the lay-up period continues into a new policy year, notice shall be given to the Association within 14 days of the start of the new policy year.

4.8 Surplus

- 4.8.1 If there is a surplus at the final closing of a policy year, the General Meeting may distribute all or part of that surplus among the members in proportion to the net annual calls paid for that policy year.
- 4.8.2 Any surplus that is not distributed shall be included in the Association's reserves.

4.9 Reserves

- 4.9.1 The Association may establish and maintain reserves as it considers appropriate, including overspill reserves in accordance with Appendix 3.
- 4.9.2 The Association may apply reserves as it considers appropriate, including the reduction of calls in any policy year.

4.10 Closing of Policy Years

- 4.10.1 The Association may decide to close a policy year at such time as it considers appropriate and taking into account unsettled claims.
- 4.10.2 When a policy year has been closed, no further premium shall be levied except for overspill calls in accordance with Appendix 3.

4.11 Payment of Premiums

- 4.11.1 Unless otherwise agreed, annual calls shall be paid in three instalments in March, July and November, except that the Association may require annual calls for less than minimum specified amounts to be paid in one instalment within 30 days of the date of the debit note.
- 4.11.2 Supplementary calls, release calls, fixed premiums and deductibles shall be paid within 30 days of the date of the debit note.
- 4.11.3 Overspill calls and, unless otherwise agreed, additional insurance premiums and all other claims which the Association may have against the member shall be paid on demand
- 4.11.4 The member shall pay the Association on demand the amount of any tax or duty relating to premiums or other sums paid or payable by the member and for which the Association is or may become liable.
- 4.11.5 The Association shall be entitled to interest on overdue payments at such rates as it may from time to time determine.

4.12 Deductibles

Unless otherwise agreed, cover shall be subject to the Association's standard deductibles as set out in Appendix 4.

4.13 Liens

The Association shall be entitled to a lien on the Vessel or, in the case of a fleet entry, any other vessel in the fleet for payments owed to it. The lien hereby granted may be enforced by the Association by way of arrest or detention of the vessel in any jurisdiction in which the vessel may call. The member hereby irrevocably confers jurisdiction and the power to order the arrest or detention of the vessel on the court adjudicating upon the Association's application and will not challenge the jurisdiction of the court to order the arrest or detention.

Part II

P&I COVER



5. CARGO LIABILITY

5.1 Cover

5.1.1 The standard insurance shall cover the member's liability for cargo loss, shortage, damage, delay or other responsibility occurring in relation to the carriage of cargo on the entered vessel.

5.2 Exceptions

However the standard insurance shall not cover liabilities, costs and expenses arising out of any of the following,

- 5.2.1 failure to arrive or late arrival of the vessel at the port of loading, other than any such liabilities, costs and expenses arising under a bill of lading already issued.
- 5.2.2 loss, shortage, damage or delay occurring prior to loading, except insofar as loss, shortage or damage occurs in the port of loading within 21 days of the date on which loading of the cargo on the vessel commences or should commence.
- 5.2.3 loss, shortage, damage or delay occurring whilst the cargo is in the custody of another carrier or during lightering operations, except insofar as lightering is approved by the Association, or occurs in port and is customary,
- 5.2.4 failure to load or delay in loading any particular cargo in the vessel, except insofar as liabilities, costs and expenses arising under a bill of lading already issued,
- 5.2.5 the issue of a bill of lading, waybill or other document containing or evidencing the contract of carriage which
 - a) is antedated or postdated,
 - b) contains a description of the cargo or its quantity or condition which the member or an officer of the vessel knows is incorrect, or
 - should be claused to show that the cargo is carried on deck and is not so claused,
- 5.2.6 carriage of cargo on terms less favourable to the member than the Hague or Hague-Visby Rules, except insofar as the contract of carriage is on less favourable terms solely because of the compulsory application of the Hamburg Rules by virtue of the place of loading or discharge,
- 5.2.7 carriage of cargo on terms which are contrary to terms required by the Association,
- 5.2.8 carriage of cargo under a contract providing for carriage partly in the entered vessel and partly by some other means of transport, except insofar as the Association approves the contract.
- 5.2.9 carriage of cash, banknotes or other forms of currency, bullion, works of art, precious or rare metals or stones, plate or other objects of a rare or precious nature, specie, bonds or other negotiable instruments,
- 5.2.10 carriage under an ad valorem bill of lading, waybill or other document containing or evidencing the contract of carriage in which a value in excess of USD 2,500 or the equivalent in any other currency per unit, piece or package or otherwise is declared or stated, where this deprives the member of the right to rely on defences or rights of limitation which would otherwise have been available to him, except insofar as liability does not exceed that sum.
- 5.2.11 deviation or departure from the contractually agreed voyage or adventure which deprives the member of the right to rely on defences or rights of limitation which would otherwise be available.

- 5.2.12 delay, except insofar as liability arises because of the application of the Hague or Hague-Visby Rules or compulsory law.
- 5.2.13 discharge of the cargo at a port or place other than the port or place provided for in the contract of carriage,
- 5.2.14 failure to discharge all the cargo on board, except insofar as the member takes all reasonable steps to discharge the cargo,
- 5.2.15 delivery of cargo carried under
 - a) a negotiable bill of lading or similar document of title without production of that bill of lading or document by the person to whom delivery is made, except where cargo has been carried on the entered vessel under the terms of a nonnegotiable bill of lading, waybill or other non-negotiable document and has been properly delivered as required by that document, and liability arises under a negotiable bill of lading or other similar document of title issued by or on behalf of a party other than the member providing for carriage partly by a means of transport other than the entered ship, or
 - b) a non-negotiable bill of lading, waybill or similar document, without production of the original document by the person to whom delivery is to be made where there is a legal obligation to produce such document, except where there is also a requirement under any other applicable law obliging the carrier under the contract of carriage contained in or evidenced by that document to deliver or relinquish custody or control of the cargo without production of such document.
- 5.2.16 loss, shortage, damage or delay occurring on land after discharge, except insofar as it occurs in the port of discharge within 21 days of discharge from the vessel,
- 5.2.17 participation in or use of an electronic trading system (as defined in Appendix 1) unless such system has been approved in writing by the Association, or
- 5.2.18 loss of or damage to cargo carried on a semi-submersible heavy-lift vessel or any other vessel designed exclusively for the carriage of heavy-lift cargo, unless the cargo is carried under a contract which has been approved by the Association (as referred to in Appendix 6).

5.3 Miscellaneous

In exceptional cases, the Board of Directors may cover, in its absolute discretion, all or part of the member's liability which would otherwise be excluded by Rule 5.2.1-5.2.8, Rules 5.2.10-5.2.18, provided that the Board is satisfied that the member took all reasonable steps to avoid the event or the circumstances giving rise to such liability.

5.4 Additional Covers

If separately agreed, the Association can arrange additional insurance to cover liability arising out of the following,

- 5.4.1 cargo loss, shortage, damage or delay occurring whilst the cargo is in the custody of another carrier pursuant to a contract of carriage for through transport or transhipment providing for carriage to be performed partly by the entered vessel (excluded from the standard insurance under Rule 5.2.3 or 5.2.8),
- 5.4.2 cargo loss, shortage or damage occurring more than 21 days before loading or more than 21 days after discharge from the vessel (excluded from the standard insurance under Rules 5.2.2 and 5.2.16),





- 5.4.3 carriage of cash, banknotes or other forms of currency, bullion, works of art, precious or rare metals or stones, plate or other objects of a rare or precious nature, specie, bonds or other negotiable instruments (excluded from the standard insurance under Rule 5.2.9).
- 5.4.4 carriage of cargo under an ad valorem bill of lading, waybill or other document containing a stated or declared value in excess of USD 2,500 per unit, piece or package (excluded from the standard insurance under Rule 5.2.10), or
- 5.4.5 unauthorised carriage of cargo on deck, deviation and some other types of departure from the contractually agreed voyage or adventure (excluded from the standard insurance under Rule 5.2.5(c) and Rule 5.2.11).

6. EXTRA CARGO HANDLING COSTS

6.1 Cover

The standard insurance shall cover the member's extra costs, and liability for extra costs, in connection with or as a consequence of handling and disposing of cargo, where such costs are necessarily, reasonably and solely incurred, as a direct result of

- 6.1.1 damage to cargo on board the entered vessel,
- 6.1.2 damage to the vessel which is of a type that would be covered under a standard hull policy, or
- 6.1.3 the consignee's rejection of cargo carried on board the vessel.

6.2 Exceptions

However the standard insurance shall not cover costs or liability for costs, which,

- 6.2.1 are claimable in general average or for which the member has a right of recourse against any other party,
- 6.2.2 result from the vessel being overloaded or improperly stowed.
- 6.2.3 are incurred in order to make the vessel seaworthy to receive the cargo,
- 6.2.4 form part of the daily running costs and expenses of the vessel,
- 6.2.5 are for work which could have been carried out by the crew or by reasonable use of the vessel and her equipment,
- 6.2.6 are in respect of packing, rebagging, sorting and other measures taken in order to comply with ordinary obligations under the contract of carriage, or
- 6.2.7 result from any of the matters referred to in Rules 5.2.1 5.2.18 (cargo liability exceptions).

7. CREW

7.1 Cover

The standard insurance for crew shall cover the member's liability for,

- 7.1.1 injury, illness or death,
- 7.1.2 hospital and medical expenses arising from injury, illness or death,
- 7.1.3 loss of or damage to personal effects,

SKULD INFORMATION: S&R 02/2011

- 7.1.4 costs of repatriation and maintenance ashore resulting from injury, illness or death, or a major casualty to the vessel which renders the vessel unseaworthy and necessitates the signing off of the crew,
- 7.1.5 the costs of the funeral or sending home of the coffin or ashes, and personal effects of a deceased crew member,
- 7.1.6 costs of repatriation incurred as a result of leave to attend a close relative, or the funeral of a close relative who has died or become seriously ill after the crew member signed on,
- 7.1.7 costs of repatriation under a statutory obligation,
- 7.1.8 wages to serving crew members or, if deceased, their dependants as a result of injury, illness or death,
- 7.1.9 compensation for loss of employment to serving crew members as a result of being signed off due to a major casualty to the vessel which renders the vessel unseaworthy, and necessitates the signing off of the crew, and
- 7.1.10 costs of providing a substitute crew member required as a result of the injury, illness or death of a crew member, or repatriation in accordance with Rule 7.1.6.

7.2 Exceptions

- However the standard insurance shall not cover liabilities, costs and expenses which arise,
- 7.2.1 under the terms of a crew contract or other agreement, unless those terms have been approved by the Association,
- 7.2.2 as a result of expiry, termination or breach by the member of a crew contract or other agreement, or expiry of the crew member's period of service, or sale of the vessel, or any other act of the member in respect of the vessel,
- 7.2.3 out of the carriage of cash or valuable objects as referred to in Rule 30.1.6, or
- 7.2.4 in respect of catering crew when the vessel is moored (otherwise than on a temporary basis) and is open to the public as a hotel, restaurant, bar or other place of entertainment.

7.3 Limitation

Cover under this Rule is subject to the limitations contained in Appendix 5 A.

7.4 Additional Cover

If separately agreed, the Association can arrange additional insurance to cover liability,

- 7.4.1 for permanently employed crew members who are ashore or supervisory crew serving on new buildings or on vessels which are about to be delivered to the member, or
- 7.4.2 arising out of the carriage of cash or other valuable objects (excluded from the standard insurance under Rule 7.2.3).



8. PASSENGERS

8.1 Cover

The standard insurance for passengers shall cover the member's liabilities, costs and expenses arising in respect of passengers carried on board the insured vessel and in respect of or consequent upon:

- 8.1.1 injury, illness or death,
- 8.1.2 hospital and medical expenses arising from injury, illness or death,
- 8.1.3 loss of or damage to baggage belonging to such a passenger.
- 8.1.4 costs of repatriation and maintenance ashore and funeral expenses resulting from injury, illness or death,
- 8.1.5 costs of return to port of embarkation or forwarding to port of destination for embarked passengers and maintenance ashore as a result of an incident stated in 8.1.8.
- 8.1.6 costs of repatriation under a deportation order,
- 8.1.7 delay arising solely because of the application of compulsory law, and
- 8.1.8 a casualty involving either collision, stranding, explosion, fire or other similar cause affecting the physical condition of the insured vessel so as to render it incapable of safe navigation to its intended destination or a threat to the life, health or safety of passengers in general.

8.2 Exceptions

- However the standard insurance shall not cover liabilities, costs and expenses
 8.2.1 which are in excess of the liability which would have arisen under the contract of carriage had the member restricted liability to the maximum extent permitted
- under the applicable law,
 8.2.2 which arise out of the carriage of cash or valuable objects as referred to in Rule
- 8.2.3 which arise in respect of carriage of the passenger by any other carrier (including by air) unless such liability arises during
 - a) repatriation under Rule 8.1.4 or return to port of embarkation or forwarding to port of destination under Rule 8.1.5, or
 - b) transportation to and from the vessel by boats in port, or
- 8.2.4 which arise under a contract in respect of an excursion from the vessel where,
 - a) a separate contract has been entered into by the passenger, whether with the member or not, or
 - b) the member has waived any rights of recourse against any party.

8.3 Limitation

Cover under this Rule is subject to the limitations contained in Appendix 5 A.

8.4 Additional Cover

If separately agreed, the Association can arrange additional insurance to cover liability arising out of the carriage of cash or other valuable objects (excluded from standard insurance under Rule 8.2.2).

9. OTHER PERSONS CARRIED ON BOARD THE VESSEL

9.1 Cover

The standard insurance shall cover the member in respect of the liabilities, costs and expenses referred to in Rules 7.1.1 - 7.1.7 arising out of the carriage on board the vessel of.

- 9.1.1 close relatives of serving crew members, or
- 9.1.2 persons other than crew or passengers provided that their presence on board
 - a) is required in connection with the routine operation of the vessel, or
 - b) has been approved by the Association.

9.2 Exceptions

However, the standard insurance shall not cover liabilities, costs and expenses which

- 9.2.1 result from any of the matters referred to in Rules 7.2.1 to 7.2.4, or
- 9.2.2 arise in respect of personnel (other than marine crew) on board the vessel (being an accommodation vessel) employed otherwise than by the member, where there has not been a contractual allocation of risks between the member and the employer of the personnel which has been approved by the Association.

9.3 Limitation

Cover under this Rule is subject to the limitations contained in Appendix 5 A.

10. PERSONS NOT CARRIED ON BOARD

10.1 Cover

The standard insurance shall cover the member in respect of liability for injury, illness or death of persons other than crew, passengers or other persons carried on board the vessel.

10.2 Exceptions

However, the standard insurance shall not cover liabilities, costs and expenses which arise,

- 10.2.1 under a contract or indemnity unless approved by the Association,
- 10.2.2 in respect of hotel and restaurant guests and other visitors and catering crew when the vessel is moored (otherwise than on a temporary basis) and is open to the public as a hotel, restaurant, bar or other place of entertainment or
- 10.2.3 in respect of personnel referred to in Rule 9.2.2.

11. STOWAWAYS, DIVERSION AND RELATED COSTS

11.1 Cover

The standard insurance shall cover the member's extra net costs (in excess of those which would have been incurred but for the presence on board or the diversion) necessarily, reasonably and solely incurred,

SKULD INFORMATION: S&R 02/2011

- 11.1.1 as a result of the presence on board the vessel of stowaways, refugees or persons saved at sea provided that the member is legally liable for such costs or they are incurred with the approval of the Association, or
- 11.1.2 for fuel, insurance, wages, stores, provisions and port charges in diverting the vessel for the purpose of.
 - a) obtaining necessary medical treatment for an injured or ill person on board the vessel.
 - embarking a necessary substitute for a crew member who is injured, ill or deceased.
 - c) landing stowaways, refugees or persons saved at sea, or
 - d) assisting in the search for or rescue of persons in distress at sea.

11.2 Exceptions

However the insurance shall not cover costs

- 11.2.1 which are recoverable from another party or insurer, or
- 11.2.2 which are incurred in respect of
 - a) the loss of freight or hire for the entered vessel, or
 - b) demurrage on, detention of or delay to the vessel.

12. COLLISION AND CONTACT LIABILITY

12.1 Cover

The standard insurance shall cover the member's liability,

- 12.1.1 a) arising out of a collision between the entered vessel and another vessel, or
 b) arising out of a contact between the entered vessel and a fixed or moveable object,
 but only if and to the extent that such liability exceeds the sum recoverable under
 the vessel's hull policies solely by reason of the fact that the liability exceeds the
 valuation of the vessel in those policies, and
- 12.1.2 for the removal of wrecks arising out of such collision or contact insofar as such liability may be covered under Rule 15.

12.2 Limitation

If, in the opinion of the Association, the entered vessel is not insured for her proper value under the hull policies, the Association's liability shall be limited to the amount by which the Association considers that the member's liability exceeds the proper value for which the vessel should have been insured, as determined by the Association in its absolute discretion.

12.3 Additional cover - collision liability (RDC)

If separately agreed, the Association shall cover, in the following proportions, the member's liability for loss of or damage to another vessel, her cargo, equipment, stores and supplies, arising out of a collision between that vessel and the entered vessel, if and to the extent that such liability is not covered under the entered vessel's hull policies,

- a) one fourth of the liability,
- b) four fourths of the liability, or
- c) any other fraction of the liability agreed with the Association.

12.4 Additional cover - contact liability (FFO)

If separately agreed, the Association shall cover the member's liability for loss of and damage to a fixed or moveable object, arising out of contact between that object and the entered vessel, provided that such liability is not covered under the vessel's hull policies.

12.5 Charterers' cover

- 12.5.1 The standard insurance for charterers shall cover the members' liability for loss and damage under Rules 12.3 and 12.4 unless otherwise agreed.
- 12.5.2 If separately agreed, the Association can arrange additional insurance to cover a charterer member's liability for the loss of or damage to the entered vessel (excluded from the standard insurance under Rule 30.2.3).

12.6 Miscellaneous

- 12.6.1 If a collision occurs involving two or more vessels belonging to the same member, the member shall be entitled to recover from the Association and the Association shall have the same rights as if the vessels had belonged to different owners.
- 12.6.2 Unless otherwise agreed between the member and the Association, if both vessels involved in a collision are to blame, then where the liability of either or both of the vessels in collision becomes limited by law, claims under Rule 12 shall be settled upon the principle of single liability, but in all other cases claims under this Rule shall be settled upon the principle of cross-liabilities, as if the owner of each vessel had been compelled to pay the owner of the other vessel such proportion of that owner's damages as may have been properly allowed in ascertaining the balance or sum payable by or to the member in consequence of the collision.

13. PROPERTY LIABILITY

13.1 Cover

The standard insurance shall cover the member's liability for loss of and damage to property which is not specified elsewhere in these Rules.

13.2 Exception

However the standard insurance shall not cover liability insured under another insurance policy.

14. POLLUTION

14.1 Cover

The standard insurance shall cover the member in respect of,

- 14.1.1 liability other than fines or other penalties arising out of the actual or threatened escape or discharge of oil or other polluting substance,
- 14.1.2 costs of measures reasonably taken for the purpose of preventing or minimising pollution or any resulting damage together with any liability for loss or damage caused by the taking of such measures.

- 14.1.3 costs incurred in order to comply with an order of any government or authority for the purpose of preventing or minimising actual or threatened pollution, and
- 14.1.4 liability and costs incurred by the member as a result of his participation in the Small Tanker Oil Pollution Indemnification Agreement (STOPIA) or the Tanker Oil Pollution Indemnification Agreement (TOPIA) or any other agreement approved by the Association for the purpose of this Rule.

14.2 Exceptions

- However the standard insurance shall not cover under this Rule or any other Rule,
- 14.2.1 costs which are required as part of the normal operation, salvage or repair of the vessel.
- 14.2.2 liabilities, losses, costs or fines in respect of actual or threatened oil pollution arising out of an incident to which the US Oil Pollution Act 1990 is applicable when the entered vessel is capable of carrying oil in bulk as cargo, unless an agreement has been entered into with the Association on the terms and conditions set out in Appendix 5, or
- 14.2.3 costs and expenses which would be recoverable in general average if the member had incorporated the York-Antwerp Rules 1994.

14.3 Limitation

Cover under this Rule and all other Rules in respect of any liabilities, losses, costs, expenses and fines arising out of or in respect of actual or threatened oil pollution is subject to the limitations contained in Appendix 5.

14.4 Additional cover

If separately agreed, the Association can arrange additional insurance to cover liability, loss, costs and fines, arising out of incidents to which the US Oil Pollution Act 1990 is applicable and which would otherwise be excluded under Rule 14.2.2, provided the member enters into an agreement on the terms set out in Appendix 5 and makes all declarations and pays all premiums required under such agreement.

14.5 TOPIA and STOPIA

A member insured in respect of a vessel which is a relevant ship as defined in the Tanker Oil Pollution Indemnification Agreement (TOPIA) shall, unless the Association otherwise agrees in writing, be a party to such Agreement for the period of entry of that vessel in the Association. A member insured in respect of a vessel which is a relevant ship as defined in the Small Tanker Oil Pollution Indemnification Agreement (STOPIA) shall, unless the Association otherwise agrees in writing, be a party to STOPIA for the period of entry of that vessel in the Association. There is no cover under this Rule or Rule 19.1.3 in respect of any event arising during a period when the member is not in compliance with such requirements unless the Association agrees in writing, or unless the Board of Directors otherwise determines.

15. WRECK REMOVAL

15.1 Cover

The standard insurance shall cover liability and costs arising out of the raising, removal, destruction or marking of the wreck of the entered vessel, or any other vessel, or their equipment, bunkers or cargo lost as a result of a casualty, insofar as the raising and other operations are compulsory by law or necessary to avoid or remove a hazard or obstruction to navigation or the costs are legally recoverable from the member.

15.2 Exceptions

However the standard insurance shall not cover liability, costs, or expenses

- 15.2.1 which are covered under the vessel's hull policies, or
- 15.2.2 where the member transfers his interest in the wreck or other property saved otherwise than by abandonment, or
- 15.2.3 arising out of removal of cargo carried on a semi-submersible heavy-lift vessel or any other vessel designed exclusively for the carriage of heavy-lift cargo, unless the cargo is carried under a contract which has been approved by the Association (as referred to in Appendix 6).

15.3 Miscellaneous

The realised value of the wreck and other property saved shall be credited to the Association

16. OBSTRUCTION

16.1 Cover

The standard insurance shall cover liability to the owners of harbours, wharves, canals or similar structures or to the owners of other vessels, arising out of the entered vessel causing an obstruction as a result of a casualty.

17. GENERAL AVERAGE CONTRIBUTIONS - CARGO

17.1 Cover

The standard insurance shall cover the member's loss in respect of general average expenditure and special charges which should be paid by the cargo interests or some other party to the maritime adventure but which are not legally recoverable solely by reason of a breach of the contract of carriage.

17.2 Exceptions

However the standard insurance shall not cover loss which results from any of the matters referred to in Rules 5.2.1-5.2.18 (cargo liability exceptions).



18. GENERAL AVERAGE CONTRIBUTIONS - HULL

18.1 Cover

The standard insurance shall cover the member's loss in respect of general average expenditure and special charges which are not recoverable under the vessel's hull policies solely by reason of the value of the vessel being assessed for contribution to general average in excess of her insured value.

18.2 Limitation

If, in the opinion of the Association, the entered vessel is not insured for her proper value under the hull policies, the Association's liability shall be limited to the amount to which the member's loss would have been limited had the vessel been insured for her proper value as determined by the Association in its absolute discretion.

19. FINES

19.1 Cover

The standard insurance shall cover the member's liability for fines or other penalties imposed.

- 19.1.1 for breach of any immigration law or regulation,
- 19.1.2 for short delivery or over-delivery of cargo, or failure to comply with regulations concerning the declaration of goods on board other than cargo, or documentation of cargo, provided the member is insured for cargo liability under Rule 5,
- 19.1.3 in respect of the accidental escape or discharge of oil or other polluting substance or threat thereof, provided the member is insured for pollution liability under Rule 14 and
- 19.1.4 for smuggling or any infringement of any customs law or customs regulation other than in relation to cargo carried on the entered vessel.

19.2 Exceptions

However the standard insurance shall not cover fines,

- 19.2.1 arising out of any of the matters referred to in Rules 5.2.1-5.2.18 (cargo liability exceptions), or
- 19.2.2 in respect of actual or threatened oil pollution arising out of an incident to which the US Oil Pollution Act 1990 is applicable when the entered vessel is capable of carrying oil in bulk as cargo, unless an agreement has been entered into with the Association on the terms and conditions set out in Appendix 5.

19.3 Limitation

Liability in respect of fines in respect of oil pollution shall be subject to the limitations contained in Appendix 5.

19.4 Discretionary cover

The Association may cover, in its absolute discretion, in whole or in part, the member's liability for fines

- 19.4.1 other than those set out in Rule 19.1, provided that the Association is satisfied that the member took all reasonable steps to prevent the infringement of the law or regulation giving rise to the fine, or
- 19.4.2 imposed on a crew member or any other party in respect of the entered vessel
 - a) provided that the member is under a legal obligation to reimburse the crew member or other party, or
 - b) to the extent that the Association determines that it was reasonable for the member to reimburse the crew member or other party.

20. CONFISCATION

20.1 Discretionary cover

The Association may cover, in whole or in part, in its absolute discretion, the member's loss in respect of the confiscation of the vessel by any legally empowered authority by reason of the infringement of any customs law or customs regulation.

20.2 Limitations

However, discretionary cover shall not be available,

- 20.2.1 for an amount exceeding the market value of the vessel without commitment at the date of the confiscation or fine.
- 20.2.2 unless the Association is satisfied that the member took all reasonable steps to prevent the infringement of the law or regulation giving rise to the confiscation or fine, and
- 20.2.3 unless, where there is a confiscation, the member has taken all reasonable steps to have the confiscation cancelled, the confiscation has lasted at least 6 months from the time the Association was notified of the confiscation, and the member has been irrevocably deprived of his interest in the vessel.

21. QUARANTINE AND DISINFECTION REQUIREMENTS

21.1 Cover

The standard insurance shall cover the member's extra costs where such costs are necessarily, reasonably and solely incurred, as a result of infectious disease on board the entered vessel, in order to comply with quarantine orders or requirements for the disinfection of the vessel, the cargo or persons aboard.

21.2 Exceptions

However the standard insurance shall not cover costs.

- 21.2.1 which form part of the daily running costs of the vessel, or
- 21.2.2 when the member orders the vessel to a port at which he knows or should know that the vessel would be quarantined, except where such orders are unavoidable.



22. SALVAGE

22.1 Cover

The standard insurance shall cover the member's liabilities, costs and expenses in respect of.

- 22.1.1 life salvage payable to third parties as a result of saving or attempting to save the life of a person on or from the entered vessel, to the extent that payment is not recoverable from hull insurers or cargo owners or underwriters.
- 22.1.2 salvage operations conducted by the entered vessel for the purpose of saving life at sea, and
- 22.1.3 special compensation payable to a salvor under Article 14 of the 1989 Salvage Convention, including when applicable by virtue of incorporation in Lloyd's Open Form of Salvage Agreement or in any other standard form salvage contract approved by the Association, or under the Special Compensation P&I Clubs Clause (SCOPIC).

22.2 Exceptions

However the standard insurance shall not cover

- 22.2.1 liabilities, costs and expenses in respect of actual or threatened oil pollution arising out of an incident to which the US Oil Pollution Act 1990 is applicable when the entered vessel is capable of carrying oil in bulk as cargo, unless an agreement has been entered into with the Association on the terms and conditions set out in Appendix 5. or
- 22.2.2 risks which are outside the scope of cover available under other Rules.

22.3 Limitation

Liabilities, costs and expenses in respect of oil pollution shall be subject to the limitations contained in Appendix 5.

22.4 Additional cover

If separately agreed, the Association can arrange cover for an owner member who is a professional salvor for liabilities, costs and expenses arising out of salvage operations.

- 22.4.1 in connection with the operation of and in respect of the member's interest in the entered vessel, or
- 22.4.2 in the absence of an entered vessel,
 - a) in respect of the escape or discharge of oil in accordance with Rule 14 (Pollution), and
 - b) for risks covered under Part II of the Rules, other than under Rule 14 [Pollution]

23. TOWAGE

23.1 Cover - towage of the entered vessel

The standard insurance shall cover the member's liability arising out of towage of the entered vessel.

- 23.1.1 under a contract entered into in the ordinary course of trading
 - a) for the purpose of entering, leaving or manoeuvring within a port when customary, or
 - b) when the vessel is entered as a barge or other vessel which is habitually towed from place to place, or
- 23.1.2 under a contract which has been approved by the Association (as referred to in Appendix 6) and upon such terms as the Association may require.

23.2 Exceptions

However the standard insurance shall not cover.

- 23.2.1 risks which are outside the scope of cover available under other Rules, or
- 23.2.2 liabilities, costs and expenses in respect of actual or threatened oil pollution arising out of an incident to which the US 0il Pollution Act 1990 is applicable when the entered vessel is capable of carrying oil in bulk as cargo, unless an agreement has been entered into with the Association on the terms and conditions set out in Appendix 5.

23.3 Limitation

Liabilities, costs and expenses in respect of oil pollution shall be subject to the limitations set out in Appendix 5.

23.4 Additional cover - towage by the entered vessel

If separately agreed, the Association can arrange additional insurance for liabilities arising out of towage by the entered vessel on contractual terms which have been approved by the Association (as referred to in Appendix 6),

- a) where the vessel is entered as a tug or other vessel engaged in towage in the ordinary course of business, or
- b) where cover has been agreed prior to the commencement of the towage.

24. MITIGATION COSTS (SUE AND LABOUR)

24.1 Cover

The standard insurance shall cover the member's extraordinary costs which are necessarily, reasonably and solely incurred, after an incident, in avoiding or minimising any liability or loss of the member covered under these Rules.

24.2 Exceptions

However the standard insurance shall not cover,

- 24.2.1 costs which are claimable in general average,
- 24.2.2 costs which result from the vessel being overloaded or improperly stowed,
- 24.2.3 costs which are incurred in order to make the vessel seaworthy to receive the cargo,
- 24.2.4 costs which form part of the daily running costs of the vessel,
- 24.2.5 costs for work which could have been carried out by the crew or by reasonable use of the vessel and her equipment, or
- 24.2.6 costs which are not either approved in advance by the Association, where it is practicable to obtain such approval, or determined by the Association in its discretion to have been reasonably incurred.



24.3 Miscellaneous

The Association shall also cover any liabilities, loss, expenses or costs which the member incurs at the direction of the Association given in writing and referring to this Rule.

25. LEGAL AND ASSOCIATED COSTS

25.1 Cover

The standard insurance shall cover legal and associated costs reasonably incurred, in agreement with the Association, in relation to claims for which the member is covered under these Rules.

25.2 Discretionary cover

The Association may cover, in whole or in part, in its absolute discretion, legal and associated costs reasonably incurred in connection with a formal enquiry into a casualty involving the entered vessel.

26. MEMBER'S OWN PROPERTY LOSS

26.1 Cover

The standard insurance shall cover loss suffered by the member as a result of the loss of or damage to his own property, to the extent that the Association would have been obliged to indemnify the member in respect of liability had the property been owned by a third party.

26.2 Exceptions

However, subject to Rule 12.6.1 (sister ships involved in a collision), the standard insurance shall not cover,

- 26.2.1 loss in respect of loss of or damage to the entered vessel, her equipment, stores, bunkers, supplies, lashings or containers, or
- 26.2.2 loss in respect of cargo owned by the member or other property which could have been insured under any other customary insurance.

26.3 Additional cover

If separately agreed, the Association as agent of the member can arrange additional insurance to cover the member's loss in respect of containers owned or leased by the member.

Part III

DEFENCE COVER



27. DEFENCE COVER

27.1 Cover

The Association shall cover the member's reasonable costs for necessary legal assistance in relation to disputes which are directly connected with the operation of the entered vessel and which are in respect of any of the following:-

- 27.1.1 the carriage of cargo,
- 27.1.2 the carriage of passengers,
- 27.1.3 contracts with port agents,
- 27.1.4 crew employment contracts,
- 27.1.5 contracts with brokers.
- 27.1.6 charterparties, and, provided agreed at the time the relevant contract is entered into, contracts of affreightment,
- 27.1.7 the use of port facilities,
- 27.1.8 loss, damage or delay to the vessel,
- 27.1.9 general average,
- 27.1.10 salvage and towage,
- 27.1.11 insurance contracts in respect of the vessel,
- 27.1.12 damage to property,
- 27.1.13 personal injuries or loss of life,
- 27.1.14 supplies to the vessel,
- 27.1.15 contracts for the repair of the vessel,
- 27.1.16 conversion, alteration, building, purchase, mortgaging or sale of the vessel (subject to Rules 27.2.8 and 27.4).

27.2 Exceptions

However, the Association shall be under no liability to reimburse members for costs:

- 27.2.1 which are incurred before the member notifies the Association of the dispute or in relation to a claim which is or can be covered under the P&I Rules,
- 27.2.2 where the member fails to carry out any of his obligations under the Rules,
- 27.2.3 where the dispute is with the Association or the Association's servants, agents or representatives, or is between joint members, co-assureds or parties with joint interests in the vessel,
- 27.2.4 where the dispute is in respect of a class action or other legal proceedings in which one or more persons sue or are sued as representatives of a group of persons with a common interest,
- 27.2.5 where the dispute arises out of damage to the vessel, to the extent that the cost of repairs exceeds the deductible or one per cent of the insured value of the vessel under the hull policies,
- 27.2.6 where the dispute arises under a management agreement,
- 27.2.7 where the vessel is not entered with the Association at the time the cause of action arises and, in the case of disputes arising out of the alteration, building, purchase, mortgage or sale of the vessel, at the time the relevant contract is entered into,
- 27.2.8 where the dispute arises under Rule 27.1.16 (the conversion, alteration, building, purchase, mortgage or sale of the vessel) unless cover has been separately agreed in writing,

- 27.2.9 where the member appoints a lawyer without the approval of the Association, or
- 27.2.10 which, in the opinion of the Association, should not be covered on any of the following grounds:
 - a) there is no reasonable relationship between the amount in dispute and the costs which are likely to be incurred,
 - b) there is no reasonable relationship between the prospects of successfully obtaining an award or judgment and the costs which are likely to be incurred,
 - there is no reasonable relationship between the prospects of successfully obtaining payment (due to the financial position of the other party or otherwise) and the costs which are likely to be incurred.
 - d) there is no reasonable relationship between the prospects of successfully defending a claim and the costs which are likely to be incurred,
 - e) the member has failed to take reasonable care in the chartering, control or management of the vessel, or the position adopted by the member is unreasonable or the member's conduct has been imprudent, improper or tainted with illegality,
 - f) the member fails to provide information or documentation which is necessary for the dispute to be properly evaluated or handled,
 - g) the member refuses to handle or settle the case in accordance with recommendations of the Association,
 - h) the member makes concessions or enters into a settlement without the approval of the Association,
 - i) the member takes steps to initiate legal action or arbitration, or makes an application to any court or arbitration tribunal, or takes any other material step in a dispute, without seeking the prior approval of the Association, or
 - j) any other reason which the Association decides, in its absolute discretion, is sufficient reason for cover not to apply

27.3 Miscellaneous

- 27.3.1 The Association may at any stage of a dispute withdraw or limit the extent of cover for any of the reasons set out in Rule 27.2. or in any of the circumstances referred to in Rules 3.1.2 and 3.1.3.
- 27.3.2 In the event that cover is withdrawn, the member is liable to reimburse the Association for any costs which the Association has previously incurred.
- 27.3.3 Where the costs of a dispute are only partly covered, the Association shall decide in its absolute discretion, on the applicable apportionment of costs.
- 27.3.4 Insofar as the member's costs are covered, the Association shall be entitled to any sum which the member recovers in respect of costs pursuant to any award, judgment or settlement agreement, and in the event that a settlement agreement does not provide, or does not provide adequately for recoverable costs, the Association shall be entitled to such sum as it considers should have been attributable to costs pursuant to such an agreement.
- 27.3.5 The Association shall decide in its absolute discretion whether a case involves one or more disputes to which a deductible applies.

27.3.6 Where a separate agreement has been reached in respect of disputes arising out of the repair, alteration, conversion, building, purchase or sale of the entered vessel (pursuant to Rule 27.2.8), any limit agreed shall apply in the aggregate to all disputes arising out of one contract or series of contracts.

27.4 Limitation

Cover under this Rule is limited to USD 5 million per dispute, except for disputes arising under Rule 27.1.16 (alteration, conversion, building, purchase, mortgage or sale of the vessel) where the limit shall be USD 300,000 per dispute unless a different limit is agreed.

Part IV

GENERAL PROVISIONS



28. CONDITIONS PRECEDENT

28.1 Disclosure and alteration of risk

The member shall,

- 28.1.1 make full and correct disclosure to the Association, before the contract of insurance is concluded, of every circumstance.
 - a) which is known to the member or any agent effecting the insurance on his behalf, or which, in the ordinary course of business, ought to be known by the member or agent, and
 - b) which would influence the Association in deciding whether and on what terms to provide cover.
- 28.1.2 make full, correct and prompt disclosure to the Association, of every change in circumstance which is or ought to be known to the member and which alters the risk covered by the Association, and
- 28.1.3 refrain from causing or agreeing, without the Association's prior approval, to any change in circumstance which alters the risk covered by the Association.
- 28.1.4 In the event of any failure to comply with any of the above requirements, the member shall not be entitled to any recovery from the Association in respect of any event occurring after the time of the failure, except insofar as there has only been a failure to comply with Rules 28.1.2 28.1.3 and the Association would have provided the cover at the same premium had the Association known of the changed circumstances prior to the conclusion of the contract.

28.2 Fraudulent, unlawful and deliberate acts

The member shall not.

- 28.2.1 make any fraudulent claim on the Association,
- 28.2.2 knowingly allow the vessel to be used for illegal purposes, or
- 28.2.3 deliberately cause or attempt to cause a casualty.
- 28.2.4 In the event of any failure to comply with any of the above requirements, the member shall not be entitled to any recovery from the Association in respect of any event occurring at or after the time of the failure.

28.3 Premiums

- 28.3.1 The member shall pay all premiums and other sums due to the Association as they fall due.
- 28.3.2 In the event of any failure to comply with the above requirement, the member shall not be entitled to any recovery from the Association in respect of any event occurring during the period when any premium or other sum was outstanding and the Association shall be entitled to cease handling all or any of the cases the Association is for the time being handling for the member.

28.4 Classification & certification

It shall be a condition precedent of the insurance cover.

- 28.4.1 that the entered vessel remains fully classed with a classification society approved by the Association,
- 28.4.2 that the vessel's classification society is not changed without the Association's prior consent, and

- 28.4.3 that the member shall maintain the validity of all statutory certificates issued by or on behalf of the state of the vessel's flag in relation to the ISM Code and ISPS Code.
- 28.4.4 In the event of any failure to comply with any of the above requirements, the member shall not be entitled to any recovery from the Association in respect of any event occurring during the period of non-compliance, except insofar as there has only been a failure to comply with Rule 28.4.2-3 where the member is a charterer and the failure was beyond the member's control.

28.5 Pay to be paid

- 28.5.1 Unless the Association shall in its absolute discretion otherwise determine, it shall be a condition precedent of the member's right to claim against the Association that the liabilities, losses, expenses or costs (which are the subject of the claim) have actually been paid or discharged by the member, joint member or co-assured and that, in the event of a liability, the liability has been discharged pursuant to:
 - a) a court order or judgment, other than a default judgment,
 - b) an award, other than a default award, of an arbitration tribunal appointed with the consent of the Association or in accordance with an arbitration agreement entered into before the event giving rise to the claim arose, or
 - c) a settlement approved by the Association.
- 28.5.2 Notwithstanding Rule 28.5.1, where a member has failed to discharge a legal liability to pay damages or compensation for personal injury, illness or death of a crew member, the Association shall discharge or pay such claim on the member's behalf directly to such crew member or dependent thereof, provided always that
 - a) the crew member or dependent has no enforceable right of recovery against any other party and would otherwise be uncompensated.
 - b) the amount payable by the Association shall under no circumstances exceed the amount which the member would otherwise have been able to recover from the Association under the Rules and the member's terms of entry, and
 - c) where the Association is under no liability to the member in respect of the claim by reason of termination under Rule 3.3.2 (b) for non-payment of premiums or other sums due to the Association, the Association shall discharge or pay the claim to the extent only that it arises from an event occurring prior to the date of termination but as agent only of the member and the member shall be liable to reimburse the Association for the full amount of such claim.

29. OTHER CONDITIONS

29.1 The member shall,

- 29.1.1 comply with any recommendations made by the Association following a survey,
- 29.1.2 comply with any directions or safety regulations issued by the Association or any applicable public authority,
- 29.1.3 comply with all rules, regulations, recommendations and requirements of the classification society,
- 29.1.4 comply with all statutory requirements of the state of the vessel's flag, relating to the construction, adaptation, condition, fitment, equipment, manning, operation, security and management of the entered vessel (including applicable requirements



- of the ISM and ISPS codes) and maintain the validity of all statutory certificates issued by or on behalf of the vessel's flag state in relation to such requirements.
- 29.1.5 provide the Association, on request, with any necessary authorisation to enable the Association to inspect and be provided with any information or documents, in the possession of the vessel's current and previous classification societies, relating to the maintenance of class.
- 29.1.6 enable the Association, at any time, to carry out a survey in accordance with Rule 35.
- 29.1.7 provide the Association with any information or documents requested by the Association in respect of the condition, manning, operation or management of the vessel.
- 29.1.8 incorporate into all contracts and indemnities any terms required by the Association (as referred to in Appendix 6), and
- 29.1.9 exclude from all contracts and indemnities any terms which are prohibited by the Association.
- 29.1.10 In the event of any failure to comply with the requirements set out in Rules 29.1.1 29.1.9, the member shall not be entitled to any recovery from the Association, except insofar as the member can prove that liabilities, losses, expenses or costs would have been incurred in any event and would have been covered by the Association if the member had complied with those requirements.

30. EXCLUSIONS

30.1 General exclusions

The insurance shall not cover liabilities, losses, expenses or costs,

- 30.1.1 which the Association excludes under Rule 35.4 (exclusion of cover following a survey),
- 30.1.2 which are recoverable under the vessel's hull policies or which, in the opinion of the Association, would have been recoverable had the vessel been properly insured for her uncommitted market value on standard terms without deductible or franchise (with the proper insured value to be determined by the Association in its absolute discretion), subject only to Rule 27 (Defence cover),
- 30.1.3 which are recoverable by the member under another insurance policy, or which would have been recoverable under another standard insurance policy but for any term in such a policy providing for deductibles or excluding or limiting liability on the grounds of double insurance,
- 30.1.4 which relate to a person performing work in the service of the entered vessel covered by social insurance or by public or private insurance required by the legislation or a collective wages agreement governing the contract of employment, or which would have been covered had such insurance been effected.
- 30.1.5 which arise out of irrecoverable debts or the insolvency of any party, including the member, or out of the fraud of agents or of an associated company or of employees of the Member acting as agent.

SKULD INFORMATION: S&R 02/2011

- 30.1.6 which arise out of the carriage of cash, bank notes or other forms of currency, bullion, works of art, precious or rare metals or stone, plate or other objects of a rare or precious nature, specie, bonds or other negotiable instruments, whether carried as cargo or as passengers' baggage or as crew effects or otherwise and whether the value is declared or not.
- 30.1.7 which result from the member knowingly sending the vessel to sea in an unseaworthy condition,
- 30.1.8 which arise in circumstances where there has been wilful misconduct (namely an act intentionally done, or a deliberate omission by the member, with knowledge that the performance or omission will probably result in injury, loss or damage, or an act done or omitted in such a way as to allow an inference of a reckless disregard of the probable consequences) on the part of the member, or
- 30.1.9 which arise out of a default judgment or default award.

30.2 Excluded losses

The insurance shall not cover, except under Rule 24 (mitigation costs) or Rule 27 (Defence), liabilities, losses, expenses or costs,

- 30.2.1 which would not have arisen but for the terms of a contract or an indemnity entered into by, or on behalf of, the member, unless the contract or indemnity is accepted or approved by the Association,
- 30.2.2 which arise in respect of salvage, towage or services in the nature of salvage provided to the entered vessel, or which arise out of salvage or towage operations performed by the member or by an entered vessel, subject to Rules 17 and 18 (general average), Rule 22 (salvage) or Rule 23 (towage),
- 30.2.3 which are incurred in respect of the loss of or damage to the entered vessel or any part of the vessel, subject only to Rule 20 (confiscation),
- 30.2.4 which are in respect of the loss of or damage to any equipment or any containers, lashing, stores or fuel on board insofar as they are owned or leased by the member or by any party associated with or under the same management as the member,
- 30.2.5 which are incurred in respect of the cost of repairs to the vessel or any charges or expenses in connection with such repairs, subject only to Rules 17 and 18 (general average),
- 30.2.6 which are incurred in respect of
 - a) the loss of freight or hire for the entered vessel, or
 - b) demurrage on, detention of or delay to the vessel, except insofar as they form part of a third party claim for liabilities in respect of cargo,
- 30.2.7 which arise out of the cancellation of a charter or other engagement of the vessel,
- 30.2.8 which are in respect of consequential loss of profit or depreciation arising from the rescue of refugees, or
- 30.2.9 which are incurred in respect of general monetary loss, consequential loss, market fluctuations, loss of market, loss of production, depreciation, loss of time, loss of opportunity, loss of profit, loss of reputation, or any similar loss, as a result of delay or the abandonment of a voyage, except where the member is legally liable to a third party for such loss and such liability is covered under Rule 5 (cargo liability) or Rule 8.1. (passengers).



30.3 Excluded operations

The insurance shall not cover, except under Rule 27 (Defence), liabilities, costs and expenses.

- 30.3.1 which are incurred in respect of a drilling or production vessel (as defined in Appendix 1) and which arise out of or during drilling or production operations,
- 30.3.2 which are incurred in connection with any claim brought against the member arising out of waste disposal activities (as defined in Appendix 1) or sub-sea activities (as defined in Appendix 1), and
- 30.3.3 which arise during the course of specialist operations (as defined in Appendix 1) as a consequence of:
 - claims brought by any party, for whose benefit the work has been performed, or by any third party (whether connected with any party for whose benefit the work has been performed or not), in respect of the specialist nature of the operation,
 - b) the failure to perform such specialist operations by the member or the fitness for purpose or quality of the member's work, products or services, or
 - c) loss of or damage to the contract work, except that this exclusion shall not apply to loss of life, injury of crew and other personnel on board the vessel, the wreck removal of the vessel, or oil pollution from the vessel insofar as covered under the Rules.

30.4 Excluded risks

The insurance shall not cover,

- 30.4.1 war risks' liabilities, costs or expenses (as defined in Appendix 1),
 - a) except that this exclusion shall not apply to liabilities, costs and expenses of a member insofar only as they are discharged by the Association on behalf of the member pursuant to a demand made under
 - (i) a FMC undertaking or
 - (ii) an undertaking given by the Association to the International Oil Pollution Compensation Fund 1992 in connection with STOPIA,
 - (iii) a CLC Certificate, or
 - (iv) a Bunkers Convention Certificate
 - to the extent such liabilities, costs and expenses are not recovered by the member under any other policy of insurance or any extension to the cover provided by the Association.
 - b) where any such guarantee, undertaking or certificate is provided by the Association on behalf of the member as a guarantor or otherwise, the member agrees that any payment by the Association thereunder in discharge of the said liabilities, costs and expenses shall, to the extent of any amount recovered under any other policy of insurance or extension to the cover provided by the Association, be by way of loan and that there shall be assigned to the Association, to the extent and on the terms that it determines in its discretion to be practicable, all the rights of the member under any other insurance and against any third party, and

- c) the member shall indemnify the Association to the extent that any payment under any such guarantee, undertaking or certificate in discharge of the said liabilities, costs and expenses is or would have been recoverable in whole or in part under a standard P&I war risk policy had the Member complied with the terms and conditions thereof.
- 30.4.2 nuclear risks' liabilities, costs and expenses (as defined in Appendix 1),
 - a) except that this exclusion shall not apply to liabilities, costs and expenses of a member insofar only as they are discharged by the Association on behalf of the member solely pursuant to a demand under a guarantee, undertaking or certificate as referred to in Rule 30.4.1(a) to the extent such liabilities, costs and expenses are not recovered by the member under any other policy of insurance or any extension to the cover provided by the Association,
 - b) where any such guarantee, undertaking or certificate is provided by the Association on behalf of the member as a guarantor or otherwise, the member agrees that any payment by the Association thereunder in discharge of the said liabilities, costs and expenses shall, to the extent of any amount recovered under any other policy of insurance or extension to the cover provided by the Association, be by way of loan and that there shall be assigned to the Association , to the extent and on the terms that it determines in its discretion to be practicable, all the rights of the member under any other insurance and against any third party, and
 - c) the member shall indemnify the Association to the extent that any payment under any such guarantee, undertaking or certificate in discharge of the said liabilities, costs and expenses is or would have been recoverable in whole or in part under a standard P&I war risk policy had the Member complied with the terms and conditions thereof.
- 30.4.3 liabilities, costs or expenses which arise out of or are consequent upon the vessel carrying contraband, blockade running or being employed in a trade or on a voyage which is unlawful or which the Board of Directors considers to be imprudent unsafe, improper or unduly hazardous,
- 30.4.4 liabilities, losses, costs or fines in respect of actual or threatened oil pollution arising out of an incident to which the US Oil Pollution Act 1990 is applicable when the entered vessel is capable of carrying oil in bulk as cargo, unless an agreement has been entered into with the Association on the terms and conditions set out in Appendix 5,
- 30.4.5 liabilities, costs or expenses which arise out of or in consequence of the member's participation in or use of an electronic trading system (as defined in Appendix 1) unless such system has been approved in writing by the Association,
- 30.4.6 liabilities, costs or expenses which arise as a result of the presence in or the actual or threatened escape or discharge from any land based dump, site, storage or disposal facility of any substance previously carried on the entered vessel as cargo, fuel, stores, waste or otherwise, save that the Association may, in its absolute discretion, decide to cover such liabilities, costs or expenses in whole or in part, or
- 30.4.7 liabilities, costs or expenses where payment by the Association or the provision of cover in respect thereof may expose the Association to the risk of being subject to a sanction, prohibition or any adverse action by a state or international organisation or competent authority.



31. BURDEN OF PROOF

- **31.1** The member shall have the burden of proving that any claim against the Association results from a risk covered under the insurance.
- **31.2** Where the claim results from a combination of a cause in respect of which liabilities, losses, expenses or costs are covered and a cause in respect of which liabilities, losses, expenses or costs are excluded, the member shall have the burden of proving that the cause covered by the insurance is the dominant cause, failing which the entire claim shall be excluded.

32. LIMITATIONS

32.1 Limitation of liability - general

- 32.1.1 The Association insures the member's liability as may ultimately be determined and fixed by law, including any laws relating to the limitation of liability, and the Association shall not be liable for any sum in excess of such legal liability.
- 32.1.2 Where a member or co-assured is entitled to limit any liability covered by the Association, there shall be no recovery in respect of such liability for more than the amount to which liability could have been limited.
- 32.1.3 Where a specified insured amount is agreed, the Association's liability shall not exceed that amount per event.
- 32.1.4 For the purposes of Rule 32, any occurrence or occurrences arising out of an event shall be treated as part of that event.
- 32.1.5 Where more than one limit applies, the Association's liability shall not exceed the lowest applicable limit.
- 32.1.6 Any liabilities of the Association arising in respect of or in connection with the actual or threatened escape or discharge of oil shall be subject to the provisions of Appendix 5.
- 32.1.7 Any liabilities of the Association under Rules 7, 8 or 9 or under any other Rule arising in respect of or in connection with liabilities to passengers, crew or other persons carried on board a vessel shall be subject to the provisions of Appendix 5 A.

32.2 Limitation of liability - owners

32.2.1 Where the member is an owner, the Association's liability shall be subject to the limitations set out in the overspill provisions in Appendix 3.

32.3 Limitation of liability - charterers

- 32.3.1 Where the member is a charterer, there shall be no recovery in excess of USD 350 million for any one vessel arising out of any one event.
- 32.3.2 Where the member and/or any other party or parties are insured under separate charterers'entries with the Association or any other insurer which participates in the Pooling Agreement and/or the International Group Re-insurance policies, the aggregate for all claims brought against the Association and other insurers shall be limited to USD 350 million for any one event for any one entered vessel. In such circumstances, the Association's liability shall be limited to that proportion of USD

350 million that the claim bears to the aggregate of all the claims recoverable from the Association and the other insurers.

32.4 Limitation of liability - joint members, co-assureds and affiliates

- 32.4.1 Where the insurance cover is extended to any joint member, co-assured or affiliate, the total liability of the Association shall in no circumstances exceed the sum that would have been recovered by the owner of the vessel had he been the sole assured.
- 32.4.2 Where the member is an owner but the member's co-assured or affiliate is a charterer, any insurance cover extended to the charterer or affiliate shall be limited to USD 350 million in the aggregate for any one vessel arising out of any one event.

32.5 Limitation of liability under separately agreed additional insurances (Appendix 7) in case of reinsurance default

In any case, the liability of the Association for any and all liabilities, losses, costs and expenses incurred by all members, co-assureds and affiliates under any one entry and which arise out of any one event, shall be limited to the sum insured in the terms of entry, provided always that to the extent the Association has reinsured the risks insured under any one entry, the Association shall (with the exception of insurance for War Risks and for Chemical, Bio-Chemical Electromagnetical Weapons and Computer Virus Risks) only be obliged to pay any amount in excess of USD 10 million or 10% of the cover limit per event whichever is the lowest as and when such funds are received by the Association from the reinsurer(s).

32.6 Limitation of liability - consortiums

Where the member is a party to a consortium agreement, there shall be no recovery in excess of USD 350 million arising out of any one event in respect of all insured parties and all vessels operating under such consortium. Where the member and any other party or parties to the consortium agreement are insured under entries with the Association or any other insurer which participates in the Pooling Agreement or the International Group Re-insurance policies, the aggregate for all claims brought against the Association and other insurers shall be limited to USD 350 million for any one event and the Association's liability shall be limited to that proportion of USD 350 million that the claim bears to the aggregate of all the claims recoverable from the Association and the other insurers.

32.7 Sanctions

The Member shall in no circumstances be entitled to recover from the Association that part of any liabilities, costs and expenses which is not recovered by the Association under the Pooling Agreement, General Excess Loss Contract or any reinsurance(s) arranged by the Association because of a shortfall in recovery from such parties or reinsurers thereunder by reason of a sanction, prohibition or adverse action against them by a state, international organisation or other competent authority or the risk thereof if payment were to be made by such parties or reinsurers. For the purpose of this Rule 32.7, "shortfall" includes any failure or delay in recovery by the Association by reason of the parties or reinsurers making payment into a designated account in compliance with the requirements of any state, international organisation or other competent authority.



33. PART TONNAGE AND PART COVER

Where a vessel is entered with the Association for less than its full tonnage or for a fraction of the member's liability, the Association shall only be liable to the member for a corresponding proportion of any liability, loss, expense or costs.

34. THE OBLIGATION TO MITIGATE

- 34.1 The member shall, upon the occurrence of any event which may give or has given rise to a claim upon the Association, take and continue to take all reasonable steps (including the preservation of any right of recourse against a third party or any right to limit liability) for the purpose of averting or minimising any liability, loss, expense or costs which may be covered by the Association.
- **34.2** If the member fails to comply with the requirements set out in Rule 34.1, the Association shall, in its absolute discretion, be entitled to refuse to cover all or part of the claim.

35. SURVEYS

- **35.1** The Association may, in its absolute discretion, conduct a survey of the condition, manning, operation or management of the entered vessel at any time.
- **35.2** Any such survey shall be carried out for the sole benefit of the Association.
- **35.3** Any such survey shall initially be carried out at the Association's expense.
- **35.4** However if the survey demonstrates that the condition of the vessel or standards of manning, operation or management are such that the vessel is unseaworthy or that the risk covered by the Association is altered,
 - a) the Association shall be entitled to exclude any risk which the Association considers, in its absolute discretion, appropriate to exclude in the light of the survey, until the condition, manning, operation and management of the vessel are confirmed by the Association to be acceptable, and
 - b) the member shall indemnify the Association in respect of the expense of the survey.
- 35.5 Where a vessel is entered subject to a survey, the Association shall be entitled to refuse to cover the member in respect of any event occurring prior to the survey, except insofar as the member can prove that liability, loss, expense or costs would have been incurred in any event and would have been covered by the Association, had such a survey taken place.
- **35.6** The member hereby authorises the Association to obtain all available information and documentation
 - a) relating to the maintenance of class of the vessel and in the possession of the classification society with which the vessel is or has been classed, and

b) relating to the issue of certificates under the ISM Code and in the possession of the administration of the vessel's flag state or an organisation recognised by the flag state for the purposes of the ISM Code.

35.7 The member or prospective member

- a) consents to and authorises the disclosure by the Association to any other insurer which participates in the Pooling Agreement any survey of a vessel undertaken on behalf of the Association either pursuant to an application for, or after entry in, the Association, and
- b) waives any rights or claims against the Association or its Managers of whatsoever nature arising in respect of or relating to the contents or opinions expressed in any survey so disclosed.

Provided always that

- such survey may only be disclosed to another insurer which participates in the Pooling Agreement when an application for an entry of such vessel is made thereto, and
- (ii) the disclosure of the survey shall be for the limited purpose only of that insurer considering an application to enter such vessel for insurance.

36. CLAIMS HANDLING

- **36.1** The Association shall be entitled,
- 36.1.1 to handle or take over the handling of any claim made against the member,
- 36.1.2 to employ or appoint lawyers, advisers, experts and other parties to handle any claim on behalf of the member,
- 36.1.3 to require the member to settle, compromise or otherwise dispose of any claim or proceedings in such manner and upon such terms as the Association considers appropriate, and
- 36.1.4 to exercise its discretion under these rules without being under any obligation to give reasons for the decision reached pursuant to the exercise of such discretion.

36.2 The member shall be obliged,

- 36.2.1 to inform the Association immediately of any claim made against the member and of any event which might give rise to a claim,
- 36.2.2 to provide the Association as soon as possible with all available information and documentation relating to any claim or to any event which might give rise to a claim,
- 36.2.3 to obtain information, undertake calculations, prepare reports and assist in all aspects of claims handling, at the member's own expense, in accordance with the Association's reasonable recommendations, requirements or directions,
- 36.2.4 to refrain from admitting liability or settling any claim without the prior approval of the Association,
- 36.2.5 to authorise the Association to employ or appoint lawyers, advisers, experts and other parties as agents on behalf of the member but under the control, direction and instructions of the Association, and



- 36.2.6 to undertake to secure prior approval from the Association for any lawyer appointed by the member, or their representatives, for any claim subject to these Rules, and to ensure that such appointed lawyer is under the control, direction and instruction of the Association.
- **36.3** If the member fails to allow the Association to exercise its rights as set out in Rule 36.1.1-36.1.3 or if the member fails to comply with the requirements set out in Rule 36.2.1-36.2.6, the Association shall, in its absolute discretion, be entitled to refuse to cover all or part of the claim.

37. TIME LIMITS

- **37.1** The member shall lose any right to compensation unless the member gives notice to the Association of any event which may give rise to a claim on the Association within six months of the member becoming aware of it.
- **37.2** In addition and in any event, any claim on the Association shall be waived and time barred unless the member commences proceedings against the Association in respect of that claim before whichever of the following dates first occurs:
 - a) six months from the receipt of written notice from the Association stating that the claim has been rejected; or
 - b) three years from the date on which the member first received sufficient information to enable a claim to be submitted to the Association: or
 - c) ten years from the date of any event which may give rise to a claim on the Association, unless the claim is dependent upon litigation or a general average adjustment still in progress at the expiry of the ten year period, in which case that period will be extended until one year after issue of the final judgment or adjustment.

38. PROVISION OF SECURITY

The Association shall be under no obligation to provide security in respect of any liability of the member or any claim against the member or the entered vessel.

39. WAIVER AND RECOVERY

39.1 In the absence of an express and specific written confirmation of cover given by the Association, no act, omission, inaction or conduct of any kind by the Association, including the provision of any security, payment of any sum or the handling of any claim or the survey of any vessel, shall be treated as a waiver of any of the Association's rights or an acceptance that the Association covers any particular claim.

- **39.3** Where the Association makes any payment in respect of any liability, loss, expense or costs, the Association shall be subrogated to any rights of recourse which the member, co-assured or affiliate may have against any third party in respect of the liability, loss, expense or costs.
- 39.4.1 Where the Association has made a payment in respect of any liability, loss, expense or costs to or on behalf of the member, any co-assured or affiliate, the whole of the recovery from a third party in respect of that liability, loss, expense or costs shall be credited and paid to the Association up to an amount equivalent to the payment made by the Association to or on behalf of the member, co-assured or affiliate, plus any costs incurred by the Association in connection with the right of recourse.
- 39.4.2 The balance (if any) of any recovery shall be credited and paid to the member up to an amount equivalent to any deductible.
- 39.4.3 The remaining balance (if any) shall be apportioned between the Association and the member in proportion to the respective amounts credited to them under Rule 39.4.1-Rule 39.4.2.
- **39.5** The Association shall, in its absolute discretion, assess the sums attributable in any recovery to costs.

40. EXCLUSION OF LIABILITY

- **40.1** The Association, its officers and employees shall not be liable in respect of any advice, survey, error or omission of the Association or any officers, employees, agents, representatives, lawyers, experts, surveyors, or other party employed or appointed by the Association, whether or not any negligence is involved.
- **40.2** The Association shall not be liable in respect of any monies which are unaccounted for either after collection by agents, representatives or lawyers employed by the Association or in respect of any monies which are entrusted to such parties for the account of the member.
- **40.3** Any recommendation, advice or findings of any lawyers, surveyors or other experts shall not, in any way, bind the Association or prejudice the Association's rights under the Rules.



41. SAVINGS OF THE MEMBER

Where the member as a consequence of an event which is covered by the Association, obtains extra revenue, saves expense or avoids liability or loss which would otherwise have been incurred and which would not have been covered by the Association, the Association may deduct and retain from the sum otherwise payable to the member, an amount which the Association considers, in its absolute discretion, equivalent to the benefit obtained by the member.

42. ASSIGNMENTS AND MORTGAGES

- **42.1** The member shall not assign or otherwise transfer any rights under these Rules or under any contract with the Association, without the prior consent of the Association.
- **42.2** The Association may, in its absolute discretion, consent to an assignment or transfer of the member's rights, subject to such terms as the Association considers appropriate.
- **42.3** An assignment or transfer without the Association's written consent shall not bind the Association.
- **42.4** In no circumstances, shall any assignee or mortgagee have a greater right than the member.

43. SET-OFF

- **43.1** The Association shall be entitled to set off any amount due from the member against any amount due to the member, any co-assured or affiliate.
- **43.2** Unless the Association agrees, a member shall only set off, against any amount due to the Association, any sum which has previously been awarded to the member from the Association by an Arbitration Tribunal appointed under Rule 47.

44. INTEREST

In no case whatsoever shall interest be paid on any amount due from the Association.

45. JOINT MEMBERS, CO-ASSUREDS, AFFILIATES AND FLEET ENTRIES

45.1 Joint members and co-assureds named on any one Certificate of Entry shall be jointly and severally liable in respect of all premiums, calls and other sums due to the Association in respect of the entered vessel.

- **45.2** Any payment by the Association to the member, or any joint member, or any coassured, or any affiliate, shall be deemed to be payment to the member and to all joint members, co-assureds and affiliates jointly and shall fully discharge the obligations of the Association in respect of that payment.
- **45.3** The contents of any communication between the Association and the member, or any joint member, or any co-assured, or any affiliate, shall be deemed to be within the knowledge of the member and all joint members, co-assureds and affiliates.
- **45.4** Any failure by the member, or any joint member, or any co-assured or any affiliate, to comply with any of the obligations under these Rules, shall be deemed to be a failure of the member and all joint members, co-assureds and affiliates.
- **45.5** Any conduct or omission by the member, or any joint member, or any co-assured or any affiliate, which would have entitled the Association to reject or reduce any claims shall be deemed to have been the failure of the member and all joint members, co-assureds and affiliates.
- **45.6** The Association shall not cover any liability, loss, expense or costs in respect of any dispute between the member and any joint member, co-assured or affiliate, or between joint members, or between co-assureds or affiliates.
- **45.7** The total liability of the Association in respect of any one event, to the member, and to any joint member, co-assured and affiliate shall not exceed such sum as would have been recoverable from the Association only by the member.
- **45.8** In the event that the total liability of the Association is less than the total sum claimed by the member and by any joint member, co-assured and affiliate, the Association shall be entitled to apportion payment in proportion to the respective amounts claimed.
- **45.9** Members or joint members named in a Certificate of Entry for one or more ships forming part of a fleet entry shall be jointly and severally liable in respect of premiums, calls and other sums due to the Association for any or all vessels in the fleet.

46. OMNIBUS RULE

- **46.1** The Association may cover, in its absolute discretion, the member's liability, loss, expense or costs which would not otherwise be covered under the Rules, to the extent that the Association considers that such cover would be appropriate and consistent with the purpose of the Association.
- **46.2** The Association, in exercising its discretion under this Rule may, in exceptional circumstances, cover liability, loss, expense or costs, which are otherwise expressly excluded.



- **46.3** The Association, in exercising its discretion under this or any other Rule may, if it wishes to do so, take into account the Rules of other Clubs of the International Group of P&I Associations or the terms and guidelines of any Pooling Agreement between any such Clubs, but the Association shall not be obliged to take into account any previous case in which the Association or any other Club exercised its discretion in a particular manner.
- **46.4** The exercise of discretion under the Rules shall be exercised by the Board of Directors, who shall delegate to the President and Chief Executive Officer and other employees of the Association or Managers insofar as the Board considers it appropriate to do so.

47. ARBITRATION AND LAW

- **47.1** Unless otherwise agreed, any dispute between the Association and any member, co-assured, affiliate, former member, or any party making a claim on the Association, shall be determined by Arbitration in Oslo.
- **47.2** Each party shall appoint one Arbitrator who shall together appoint a chairman of the arbitration tribunal, although if the two arbitrators fail to reach agreement on the appointment of a chairman, the chairman shall be appointed by the Chief Justice of the Oslo City Court.
- **47.3** The Rules and any arbitration proceedings shall be governed by Norwegian Law, except that the Insurance Contracts Act of 1989 shall not apply.

APPENDICES



APPENDIX 1 DEFINITIONS

In the Rules the following words shall have the following meanings:-

Affiliate

A person other than a co-assured who is affiliated to or associated with the member and to whom the Association has agreed (subject to restrictions) to extend the cover afforded to the member.

Ante dated or post dated bill of lading, etc.

A bill of lading, waybill or other document containing or evidencing the contract of carriage which records the loading or shipment or receipt for shipment on a date prior or subsequent to the date on which the cargo was in fact loaded, shipped or received.

Billion

One thousand million

Bunkers Convention Certificate

A certificate issued by the Association in compliance with Article 7 of the International Convention on Civil Liability for Bunker Oil Pollution Damage 2001

Charterer

A charterer (other than a bareboat or demise charterer) of an entered vessel, or a charterer of part of an entered vessel provided that:-

- (i) in the determination of the Association that party charters part of the entered vessel for the purposes of a regular liner service provided by him; or
- (ii) in addition to his entry with the Association as a charterer of part of the entered vessel, he has a separate entry with the Association or other insurer which is party to the International Group's Pooling Agreement as an owner, or as a charterer of an entered vessel.

A charterer cannot become a member where an owner or another charterer is insured under the same entry.

Chemical, Bio-Chemical Electromagnetical Weapons and Computer Virus Risks

Loss, damage, liability or expense directly or indirectly caused by or contributed to by or arising from

- (i) any chemical, biological, bio-chemical or electromagnetic weapon
- (ii) the use or operation, as a means of inflicting harm, of any computer virus.

CLC Certificate

A certificate issued by the Association in compliance with Article VII of the International Conventions on Civil Liability for Oil Pollution Damage 1969 or 1992 or any amendments thereof.

Co-assured

A party, other than the member, who is named on the Certificate of Entry, to whom the Association has agreed (subject to restrictions) to extend the cover afforded to the member.

Consortium Agreement

Any arrangement under which a member agrees with other parties to the reciprocal exchange or sharing of cargo space on the entered vessel and on a vessel or vessels other than the entered vessel,

Crew

Persons, including the master, contractually obliged to serve on an entered vessel (except persons engaged only for nominal pay), including substitutes for such persons and including such persons while proceeding to or from the vessel.

Drilling or production vessel

A drilling vessel or barge or any other vessel or barge employed to carry out drilling or production operations in connection with oil or gas exploration or production, including any accommodation unit moored or positioned on site as an integral part of any such operations.

A vessel shall be deemed to be carrying out production operations if (inter alia) it is a storage tanker or other vessel engaged in the storage of oil, and either:

- a) the oil is transferred directly from a producing well to the storage vessel; or
- b) the storage vessel has oil and gas separation equipment on board and gas is being separated from oil whilst on board the storage vessel other than by natural venting.

Electronic trading system

Any system which replaces or is intended to replace paper documents used for the sale of goods and/or their carriage by sea or partly by sea and other means of transport and which

- a) are documents of title, or
- b) entitle the holder to delivery or possession of the goods referred to in such documents, or
- c) evidence a contract of carriage under which the rights and obligations of either of the contracting parties may be transferred to a third party.

For the purpose of this definition, a "document" shall mean anything in which information of any description is recorded, including, but not limited to, computer or other electronically generated information.

Event

Any event, including any occurrence or occurrences arising out of any such event unless the Association elects to treat each occurrence as a separate event. An event shall be deemed to have taken place at the time of the first occurrence which results in a claim or claims.

FMC Undertaking

A guarantee or other undertaking given by the Association to the Federal Maritime Commission under Section 2 of US Public Law 89-777.

Fleet entry

The entry of more than one vessel by one or more members on the basis that those vessels shall be treated together as a fleet for underwriting purposes.



General Excess Loss Contract

The General Excess Loss Contract effected by the parties to the Pooling Agreement.

Hague Rules

The International Convention for the Unification of Certain Rules Relating to Bills of Lading signed at Brussels on 25 August 1924.

Hague-Visby Rules

Hague Rules as amended by the Protocol signed at Brussels on 23 February 1968.

Hamburg Rules

The United Nations Convention on the Carriage of Goods by Sea done at Hamburg on 31 March 1978.

Hull policies

Policies effected on the hull and machinery of a vessel, including any excess liability policy.

ISM Code

International Safety Management Code.

ISPS Code

The International Ship and Port Facility Security Code.

Nuclear Risks liabilities, costs and expenses

Liabilities, costs and expenses directly or indirectly caused by or contributed to by or arising from

- a) Ionising raditions from or contamination by radioactivity from any nuclear fuel or from any nuclear waste or from the combustion of nuclear fuel,
- b) the radioactive, toxic, explosive or other hazardous or contaminating properties of any nuclear installation, reactor or other nuclear assembly or nuclear component thereof,
- c) any weapon or device employing nuclear fission and/or fusion or other like reaction or radioactive force or matter, or
- d) the radioactive, toxic, explosive or other hazardous or contaminating properties of any radioactive matter

other than liabilities, losses, costs or expenses arising out of carriage of "excepted matter" (as defined in the Nuclear Installations Act 1965 of the United Kingdom or any regulations made thereunder) as cargo in an entered vessel.

Owner

Any owner, owner in partnership or owner holding separate shares in severalty, part owner, trustee or bareboat or demise charterer of any entered vessel, any manager or operator having control of the operation and employment of an entered vessel (being such control as is customarily exercised by a shipowner), and any other person in possession and control of any entered vessel.

Passenger

Any person carried on board an entered vessel pursuant to a ticket of passage.

Person

A person shall include, unless the context otherwise requires, any individual or any body corporate or unincorporate.

Personal effects

Personal property including clothes, documents, navigation and other technical instruments and tools but excluding valuables and any other articles which, in the opinion of the Association, are not reasonably required by a crew member.

Policy year

A year from noon GMT on 20 February to immediately prior to noon GMT on the next following 20 February.

Pooling Agreement

An Agreement, to which the Association is a party, between certain Protection and Indemnity Associations dated 20 February 2002 and any addendum to, or variation or replacement of such agreement.

Specialist operations

Operations including and not limited to dredging, blasting, pile-driving, well-stimulation, cable or pipelaying, construction, installation, removal or maintenance work, core sampling, depositing of spoil, professional oil spill response or professional oil spill response training and tank cleaning (otherwise than on the entered vessel) but excluding fire fighting.

STOPIA

Small Tanker Oil Pollution Indemnification Agreement including STOPIA 2006 or any other amendment thereof.

Sub-Sea activities

The operation by the member of submarines, mini-submarines or diving bells, or the activities of professional or commercial divers where the member is responsible for such activities (other than activities arising out of salvage operations being conducted by an entered vessel where the divers form part of the crew of that entered vessel (or of diving bells or other similar equipment or craft operating from the entered vessel) and where the member is responsible for the activities of such divers).

TOPIA

Tanker Oil Pollution Indemnification Agreement 2006 or any amendment thereof.

Vessel

Any ship, boat, hydrofoil, hovercraft or other description of vessel (including a lighter, barge or similar vessel howsoever propelled but excluding a fixed platform or fixed rig) used or intended to be used for any purpose whatsoever in navigation or otherwise on, under, over or in water or any part of such ship, boat, hydrofoil, hovercraft or other description of vessel or any proportion of the tonnage thereof or any share therein.



War Risks' liabilities, costs or expenses

Liabilities, costs or expenses (irrespective of whether a contributory cause of the same being incurred was any neglect on the part of the member or his servants or agents) when the loss or damage, injury, illness or death or other accident in respect of which such liability arises or cost or expense is incurred, was caused by:

- a) war, civil war, revolution, rebellion, insurrection or civil strife arising therefrom, or any hostile act by or against a belligerent power, or any act of terrorism (provided that, in the event of any dispute as to whether or not, for the purposes of this paragraph (a), an act constitutes an act of terrorism, the decision of the directors of the Association shall be final):
- b) capture, seizure, arrest, restraint or detainment (barratry and piracy excepted) and the consequences thereof or any attempt thereat;
- mines, torpedoes, bombs, rockets, shells, explosives or other similar weapons of war, provided that this exclusion shall not apply to liabilities, costs and expenses which arise solely by reason of
- (i) the transport of any such weapons whether on board or not, or
- (ii) the use of any such weapons, either as a result of government order or with the agreement in writing of the Association where the reason for such use was the avoidance or mitigation of liabilities, costs or expenses which would otherwise give rise to a claim covered under these Rules.

Waste disposal activities

Waste incineration or disposal operations carried out by the entered vessel (other than any such operations carried out as an incidental part of other commercial activities not being specialist operations).

Wilful misconduct

An act intentionally done, or a deliberate omission by the member, with knowledge that the performance or omission will probably result in injury, loss or damage, or an act done or omitted in such a way as to allow an inference of a reckless disregard of the probable consequences.

APPENDIX 2 TERMS OF ENTRY FOR CO-ASSUREDS AND AFFILIATES

1. Co-assurance

- 1.1 The Association may agree to extend the cover afforded to the member to a coassured named in the Certificate of Entry who is
- a) a person interested in the operation, management or manning of the entered vessel,
- b) the holding company or the beneficial owner of the member or of any co-assured referred to in (a), or
- c) the mortgagee of the vessel.
- 1.2 The cover afforded to a co-assured referred to in paragraph 1.1 shall extend only to risks, liabilities and expense arising out of operations and/or activities customarily carried on by or at the risk and responsibility of shipowners (or charterers, if the member is a charterer) and which are within the scope of cover afforded by the Rules and any special terms set out in the Certificate of Entry.

2. Charterer's co-assurance

- 2.1 The Association may agree to extend the cover afforded to the member to a coassured who is named in the Certificate of Entry and who is a charterer of the entered vessel and also being affiliated to or associated with the member insured under the same entry.
- 2.2 A co-assured referred to in paragraph 2.1 shall only be covered for the risks, liabilities and expenses in respect of which the affiliated or associated member has cover and shall be limited by Rule 32.4.2 (charterers' limitation of liability).

3. Contractor's co-assurance

- 3.1 The Association may agree to extend the cover afforded to the member to a coassured named in the Certificate of Entry who is a person (a "contractor") who has entered into a contract (a "knock for knock" contract as customarily used in the offshore activity) with the member for the provision of services to or by the vessel, and any of the contractor's affiliates, other contractors and co-licensees and its and their sub-contractors and any of their personnel, provided that the contract has been approved by the Association and the contract is on terms no less favourable to the insured owner than that each party shall be responsible for loss of or damage to, or injury or death of, its own property and personnel and property and personnel of its affiliates and other contractors (and in the case of off-shore operators, co-licensees) and its and their sub-contractors, irrespective of any fault or neglect of that party or its affiliates, contractors or co-licensees or its or their sub-contractors or of any of their personnel.
- 3.2 A co-assured referred to in paragraph 3.1 above shall only be covered for liabilities, costs and expenses which are to be borne by the member under the terms of the contract and which would, if borne by the member, be recoverable by the member from the Association.

4. Affiliate cover

4.1 The Association may agree to extend the cover afforded by the Association to the member to any affiliate who is affiliated to or associated with the member but who is not specifically named in the Certificate of Entry.



4.2 The cover afforded to a person referred to in paragraph 4.1 shall be restricted to cover where the Certificate of Entry expressly extends cover generally to affiliates, but without specifically naming the affiliate or affiliates. The cover shall also be restricted only to cover a claim in respect whereof the member named in the Certificate of Entry is insured by the Association, is made or enforced through a person or company affiliated or associated with the member. The Association shall then, if so requested by the member, indemnify such person or company against any loss which as a consequence thereof such person or company shall have incurred in that capacity provided always that nothing herein contained shall be construed as extending to any amount which would not have been recoverable from the Association by the member had such claim been made or enforced against him. Once the Association has made such indemnification under the above cover, it shall not be under any further liability and shall not make any further payment to any person or company whatsoever. including the member, in respect of the loss or damage in respect of which the claim was brought.

5. Miscellaneous

Where cover is extended as set out in paragraphs 1-4 above, conduct of any one of the parties insured which is sufficient to bar that insured's right under the policy shall bar the rights of recovery of all the said insured.

APPENDIX 3 OVERSPILL CLAIMS AND CALLS

1. Interpretation

1.1 In this Appendix the following words and expressions shall have the following meanings:

"Convention Limit"

in respect of a vessel, the limit of liability of the shipowner of that vessel for claims (other than claims for loss of life or personal injury) at the Overspill Claim Date, calculated in accordance with Article 6 paragraph 1(b) of the International Convention on Limitation of Liability for Maritime Claims 1976 (the "Convention") and converted from Special Drawing Rights into United States Dollars at the rate of exchange conclusively certified by the Association as being the rate prevailing on the Overspill Claim Date provided that,

- a) where a vessel is entered for a proportion (the "relevant proportion") of its tonnage only, the Convention Limit shall be the relevant proportion of the limit of liability calculated and converted as aforesaid,
- b) each vessel shall be deemed to be a seagoing ship to which the Convention applies, notwithstanding any provision in the Convention to the contrary.

"Group General Excess Loss Contract"

the excess of loss reinsurance policies effected by the parties to the Pooling Agreement.

"Group Reinsurance Limit"

the amount of the smallest claim (other than any claim arising in respect of oil pollution) incurred by the Association or by any other party to the Pooling Agreement which would exhaust the largest limit for any type of claim (other than a claim arising in respect of oil pollution) from time to time imposed in the Group General Excess Loss Contract.

"Overspill Call"

a call levied by the Association pursuant to paragraph 5 for the purpose of providing funds to pay part of an Overspill Claim.

"Overspill Claim"

that part (if any) of a claim (other than a claim arising in respect of oil pollution) incurred by the Association or by any other party to the Pooling Agreement under the terms of entry of a vessel which exceeds or may exceed the Group Reinsurance Limit.

"Overspill Claim Date"

in relation to any Overspill Call, the time and date on which there occurred the incident or event giving rise to the Overspill Claim in respect of which the Overspill Call is made or, if the policy year in which such incident or event occurred has been closed in accordance with the provisions of paragraphs 6.1 and 6.2, noon GMT on 20th August of the policy year in respect of which the Association makes a declaration under paragraph 6.3.

SKULD INFORMATION: S&R 02/2011

"Pooling Agreement"

the agreement, to which the Association is a party, between certain protection and indemnity associations dated 20 February 1998 and any addendum to or variation or replacement of the said agreement.

- 1.2 All claims (other than claims or parts of claims arising in respect of oil pollution) incurred by the Association or by any other party to the Pooling Agreement under the entry of any one vessel arising from any one incident or event including any claim in respect of liability for the removal or non-removal of any wreck shall be treated for the purposes of this Appendix as if they were one claim.
- 1.3 Any reference to a claim incurred by the Association or by any other party to the Pooling Agreement shall be deemed to include the costs and expenses associated therewith.

2. Recoverability of Overspill Claims

- 2.1 Without prejudice to any other applicable limit, any Overspill Claim incurred by the Association shall not be recoverable from the Association in excess of the aggregate of
 - a) that part of the Overspill Claim which is eligible for pooling under the Pooling Agreement but which, under the terms of the Pooling Agreement, is to be borne by the Association; and
 - b) the maximum amount that the Association is able to recover from the other parties to the Pooling Agreement as their contributions to the Overspill Claim.
- 2.2 The aggregate amount referred to in paragraph 2.1 shall be reduced to the extent that the Association can evidence
 - a) that costs have been properly incurred by it in collecting or seeking to collect
 - (i) Overspill Calls levied to provide funds to pay that part of the Overspill Claim referred to in paragraph 2.1 (a), or
 - (ii) the amount referred to in paragraph 2.1 (b); or
 - b) that it is unable to collect an amount equal to that part of the Overspill Claim referred to in paragraph 2.1 (a) which it had intended to pay out of the levy of Overspill Calls because any Overspill Calls so levied, or parts thereof, are not economically recoverable, provided that if, due to a change in circumstances, such amounts subsequently become economically recoverable, the aggregate amount referred to in paragraph 2.1 shall be reinstated to that extent.
- 2.3 In evidencing the matters referred to in paragraph 2.2 (b) the Association shall be required to show that
 - a) it has levied Overspill Calls in respect of the Overspill Claim referred to in paragraph 2.1 on all members entered in the Association on the Overspill Claim Date in accordance with and in the maximum amounts permitted under paragraph 5;
 and
 - it has levied those Overspill Calls in a timely manner, has not released or otherwise waived a member's obligation to pay those calls and has taken all reasonable steps to recover those calls.

3. Payment of Overspill Claims

3.1 The funds required to pay any Overspill Claim incurred by the Association shall be provided

- a) from such sums as the Association is able to recover from the other parties to the Pooling Agreement as their contributions to the Overspill Claim,
- b) from such sums as the Association is able to recover from any special insurance which may, in the discretion of the Association, have been effected to protect the Association against the risk of payments of Overspill Claims, and
- c) from such proportion as the Association in its discretion determines of any sums standing to the credit of such Overspill Reserves as the Association may in its discretion have established, and
- d) by levying one or more Overspill Calls in accordance with paragraph 5, irrespective of whether the Association has sought to recover or has recovered all or any of the sums referred to in paragraph 3.1 (b) but provided the Association shall first have made a determination in accordance with paragraph 3.1 (c), and
- e) from any interest accruing to the Association on any funds provided as aforesaid.
- 3.2 The funds required to pay such proportion of any Overspill Claim incurred by any other party to the Pooling Agreement which the Association is liable to contribute under the terms of the Pooling Agreement shall be provided in the manner specified in paragraph 3.1 (b)-(e).
- 3.3 To the extent that the Association intends to provide funds required to pay any Overspill Claim incurred by it in the manner specified in paragraph 3.1 (d), the Association shall only be required to pay such Overspill Claim as and when such funds are received by it, provided that it can show from time to time that, in seeking to collect such funds, it has taken the steps referred to in paragraph 2.3 (a) and (b).

4. Overspill Claims - Expert Determinations

- 4.1 Any of the issues referred to in paragraph 4.2 on which the Association and a member cannot agree shall be referred to a panel (the "Panel") constituted in accordance with arrangements established in the Pooling Agreement which, acting as a body of experts and not as an arbitration tribunal, shall determine the issue.
- 4.2 This paragraph 4 shall apply to any issue of whether, for the purpose of applying any of paragraphs 2.2, 2.3 and 3.3 in relation to any Overspill Claim (the "relevant Overspill Claim"),
 - a) costs have been properly incurred in collecting or seeking to collect Overspill Calls. or
 - b) any Overspill Call or part thereof is economically recoverable, or
 - c) in seeking to collect the funds referred to in paragraph 3.3, the Association has taken the steps referred to in that paragraph.
- 4.3 If the Panel has not been constituted at a time when a member wishes to refer an issue to it, the Association shall, on request by the member, give a direction for the constitution of the Panel as required under the Pooling Agreement.
- 4.4 The Association may (and, on the direction of the member, shall) give such direction as is required under the Pooling Agreement for the formal instruction of the Panel to investigate any issues and to give its determination as soon as reasonably practicable.
- 4.5 The Panel shall in its discretion decide what information, documents, evidence and submissions it requires in order to determine an issue and how to obtain these, and the Association and the member shall co-operate fully with the Panel.

CKIII D INIEODMATIONI, C& D 02/2011

- 4.6 In determining any issue referred to it under this paragraph 4 the Panel shall endeavour to follow the same procedures as it follows in determining issues arising in respect of the relevant Overspill Claim which are referred to it under the Pooling Agreement.
- 4.7 In determining an issue the members of the Panel
 - a) shall rely on their own knowledge and expertise, and
 - b) may rely on any information documents, evidence or submission provided to it by the Association or the member as the Panel sees fit.
- 4.8 If the three members of the Panel cannot agree on any matter, the view of the majority shall prevail.
- 4.9 The Panel shall not be required to give reasons for any determination.
- 4.10 The Panel's determination shall be final and binding upon the Association and the member (subject only to paragraph 4.11) and there shall be no right of appeal from such determination.
- 4.11 If the Panel makes a determination on an issue referred to in paragraph 4.2 (b) or (c) the Association or the member may refer the issue back to the Panel, notwithstanding paragraph 4.10, if it considers that the position has materially changed since the Panel made its determination.
- 4.12 The costs of the Panel shall be paid by the Association.
- 4.13 Costs, indemnities and other sums payable to the Panel by the Association in relation to any Overspill Claim, whether the reference to the Panel has been made under this paragraph 4 or under the Pooling Agreement, shall be deemed to be costs properly incurred by the Association in respect of that Overspill Claim for the purposes specified in paragraph 2.2 (a).

5. Levying of Overspill Calls

- 5.1 If
 - a) The Association shall at any time determine that funds are or may in future be required to pay part of an Overspill Claim (whether incurred by the Association or by any other party to the Pooling Agreement), and
 - b) the Association shall have made a declaration under 6.1 or 6.3 that a policy year shall remain open for the purpose of levying an Overspill Call or Calls in respect of that Overspill Claim,
 - the Association in its discretion, at any time or times after such declaration has been made, may levy one or more Overspill Calls in respect of that Overspill Claim in accordance with paragraph 5.2.
- 5.2 The Association shall levy any such Overspill Call
 - a) on all members entered in the Association on the Overspill Claim Date in respect of vessels entered by them at that time, notwithstanding the fact that, if the Overspill Claim Date shall be in a policy year in respect of which the Association has made a declaration under paragraph 6.3, any such vessel may not have been entered in the Association at the time the relevant incident or occurrence occurred, and
 - b) at such percentage of the Convention Limit of each such vessel as the Association in its discretion shall decide

- 5.3 An Overspill Call shall not be levied in respect of any vessel entered on the Overspill Claim Date with an overall limit of cover equal to or less than the Group Reinsurance Limit.
- 5.4 The Association shall not levy on any member in respect of the entry of any one vessel an Overspill Call or Calls in respect of any one Overspill Claim exceeding in the aggregate two and one half per cent (2.5%) of the Convention Limit of that vessel.
- 5.5 If at any time after the levying of an Overspill Call upon the members entered in the Association in any policy year, it shall appear to the Association that the whole of such Overspill Call is unlikely to be required to meet the Overspill Claim in respect of which such Overspill Call was levied, the Association may decide to dispose of any excess which in the opinion of the Association is not so required in one or both of the following ways:
 - a) by transferring the excess or any part thereof to the Overspill Reserve in accordance with paragraph 8; or
 - b) by returning the excess or any part thereof to those members who have paid that Overspill Call in proportion to the payments made by them.

6. Closing of Policy Years for Overspill Calls

- 6.1 If at any time prior to the expiry of a period of thirty-six months from the commencement of a policy year (the "relevant policy year"), any of the parties to the Pooling Agreement sends a notice (an "Overspill Notice") in accordance with the Pooling Agreement that an incident or occurrence has occurred in the relevant policy year which has given or at any time may give rise to an Overspill Claim, the Association shall as soon as practicable declare that the relevant policy year shall remain open for the purpose of levying an Overspill Call or Calls in respect of that claim and the relevant policy year shall not be closed for the purpose of making an Overspill Call or Calls in respect of that claim until such date as the Association shall determine.
- 6.2 If at the expiry of the period of thirty-six months provided for in paragraph 6.1, no Overspill Notice as therein provided for has been sent, the relevant policy year shall be closed automatically for the purpose of levying Overspill Calls only, whether or not closed for any other purposes, such closure to have effect from the date falling thirtysix months after the commencement of the relevant policy year.
- 6.3 If at any time after a policy year has been closed in accordance with the provisions of paragraphs 6.1 and 6.2, any of the parties to the Pooling Agreement sends an Overspill Notice in accordance with the Pooling Agreement that an incident or occurrence has occurred in that Policy Year which has given or at any time may give rise to an Overspill Claim, the Association shall as soon as practicable declare that the earliest subsequent open policy year (not being a policy year in respect of which the Association has already made a declaration in accordance with paragraph 6.1 or 6.3) shall remain open for the purpose of levying an Overspill Call or Calls in respect of that claim and such open policy year shall not be closed for the purpose of making an Overspill Call or Calls in respect of that claim until such date as the Association shall determine.
- 6.4 A policy year shall not be closed for the purpose of levying Overspill Calls save in accordance with this paragraph 6.



7. Security for Overspill Calls on Termination or Cessor

7.1 I

- a) the Association makes a declaration in accordance with paragraphs 6.1 or 6.3 that a policy year shall remain open for the purpose of levying an Overspill Call or Calls. and
- b) a member who is liable to pay any such Overspill Call or Calls as may be levied by the Association in accordance with paragraph 5 ceases or has ceased to be insured by the Association for any reason, or the Association determines that the insurance of any such member may cease,

the Association may require such member to provide to the Association by such date as the Association may determine (the "due date") a guarantee or other security in respect of the member's estimated future liability for such Overspill Call or Calls, such guarantee or other security to be in such form and amount (the "guarantee amount") and upon such terms as the Association in its discretion may deem to be appropriate in the circumstances.

- 7.2 Unless and until such guarantee or other security as is required by the Association has been provided by the member, the member shall not be entitled to recovery from the Association of any claims whatsoever and whensoever arising in respect of any and all vessels entered in the Association for any policy year by him or on his behalf.
- 7.3 If such guarantee or other security is not provided by the member to the Association by the due date, a sum equal to the guarantee amount shall be due and payable by the member to the Association on the due date, and shall be retained by the Association as a security deposit on such terms as the Association in its discretion may deem to be appropriate in the circumstances.
- 7.4 The provision of a guarantee or other security as required by the Association (including a payment in accordance with paragraph 7.3) shall in no way restrict or limit the member's liability to pay such Overspill Call or Calls as may be levied by the Association in accordance with paragraph 5.

8. Overspill Reserve

- 8.1 The Association may, in its discretion, establish and maintain a reserve (an "Overspill Reserve") to provide a source of funds which may be applied towards meeting any Overspill Claim or Claims.
- 8.2 Funds to be applied to the Overspill Reserve may be raised in any of the following ways:
 - a) the Association, when deciding on the rate of any advance call or supplementary call for any policy year, may resolve that any specified amount or proportion of such call shall be transferred to and applied for the purposes of the Overspill Reserve;
 - the Association may on the closing of any policy year or at any time or times thereafter resolve that any specified amount or proportion of the funds standing to the credit of that policy year shall be transferred to and applied for the purposes of the Overspill Reserve;
 - c) the Association may transfer to the Overspill Reserve any balance of an Overspill Call not required to satisfy the Overspill Claim in respect of which it was levied, as contemplated in paragraph 5.5 (a).

APPENDIX 4 DEDUCTIBLES

1. Deductibles - P&I cover

Unless otherwise agreed, compensation shall be subject to the following deductibles per category:

1.1 For Rules other than those referred to in paragraphs 1.2 to 1.6:

USD 7,500

1.2 Rules 5 (cargo liability), 6 (extra cargo handling costs), 17 - 18 (general average) and 19.1.2 - 3 (cargo fines) applicable per cargo voyage:

USD 12,500 For vessels under 2500 GT, USD 7,500

The deductibles applicable to cargo claims under the Rules referred to in this paragraph (whether for standard or separately agreed amounts) shall be doubled when the loss or liability is due to water damage resulting from leaky cargo hatches or tank hatches on account of inadequate maintenance.

1.3 Claims under Rules 7, (crew), 8 (passengers), 9 (other persons carried on board) and 10 (persons not carried on board):

USD 7,500

1.4 Rules 14 (pollution), 19.1.3 (pollution fines) and 22.1.3 (salvage - special compensation):

USD 7,500

1.5 Rules 12 (collision and contact liability) and 13 (property liability):

Rule 12.1 (standard cover)

USD 42,500

Rule 12.3 (additional cover) RDC (4/4): USD 42,500

RDC (1/4): USD 42,500 Rule 12.4 (additional cover)

FFO: USD 42,500 Rule 13 (property liability)

USD 12,500

1.6 Rules 25 (legal and associated costs such as correspondents' fees) and 24 (mitigation costs) shall be subject to the deductible applicable to the claim in respect of which the costs have been incurred.

2. Deductibles - defence cover

Unless otherwise agreed, cover under Rule 27 (Defence cover) shall be subject to the following deductible per dispute: 25% of the total costs with a minimum of USD 5.000.

3. Miscellaneous

- 3.1 Unless otherwise agreed or stipulated otherwise in these Rules, the applicable deductibles shall apply to any one event (as defined in Appendix 1) any one vessel.
- 3.2 Where deductibles apply per port of call and the vessel is staying for an unbroken period in the same port, the deductibles shall apply for a two-month period and for coasters and vessels engaged in local trade, for a one-month period.
- 3.3 The deductible for claims in respect of cargo referred to in paragraph 1.2, and crew personal effects (Rule 7.1.3) and passengers' baggage (Rule 8.1.3) shall apply per any one voyage.
- 3.4 The deductibles may be paid in any currency at the USD rate at the date of payment.
- 3.5 If a claim under Rule 11.1.2(a) or (b) arises out of the same event as a claim under Rules 7, 8 or 9, a separate deductible shall not apply to the costs under Rule 11(a) or (b).

APPENDIX 5 OIL POLLUTION

1. US Trading

- 1.1 US oil pollution cover
 - The exclusion for liability, loss, costs and fines in respect of actual or threatened oil pollution arising out of an incident to which the US Oil Pollution Act of 1990 ("OPA 90") is applicable shall apply to all assured under any entry which insures the registered owner of a vessel which is capable of carrying oil in bulk as cargo, unless the member agrees to the following terms and conditions:
- 1.2 The member shall declare quarterly in arrears at the end of each quarter ending 20 May, 20 August, 20 November and 20 February whether or not the vessel has performed any voyage (a "relevant voyage") carrying oil as cargo to or from any port or place in the United States of America or within the exclusive economic zone ("EEZ") as defined in OPA 90 and any amendments thereto and, if so, the number of voyages, and for each voyage, the nature and (in the case of cargoes of persistent oil carried by parcel tankers) quantity of the cargoes, the port(s) or place(s) of loading, discharging or transferring such cargoes and the date(s) of such loading, discharging or transfer.
- 1.3 If the vessel makes a relevant voyage involving loading or discharging persistent oil as cargo at any port or place in the US or within the EEZ, the member shall pay an additional premium to be determined by the Association.
- 1.4 The member shall pay such additional premium on or before the date specified in the debit note issued by the Association in accordance with the declarations made under paragraph 1.2 above.
- 1.5 In the event that the member fails for any reason to make a declaration (whether or not any relevant voyage has been performed) within two months of the quarter dates specified in paragraph 1.2 above, the exclusion for oil pollution set out in Rules 14.2.2 and 30.4.4 will again come into effect for the period covered by such declaration.
- 1.6 In the event that any declaration made by the member or on his behalf pursuant to paragraph 1.2 above is in any material respect inaccurate, the Association shall be entitled to refuse cover for the period covered by such declaration.
- 1.7 In the event that the member makes no declaration, or a late or inaccurate declaration, the Association shall be entitled to refuse cover in respect of all periods subsequent to the period covered by such declaration.

2. Limitation - owners' entries

- 2.1 Cover in respect of all claims arising in respect of the actual or threatened escape or discharge of oil, including claims by charterers co-assured under an owners' entry, shall be subject to a limit of USD 1 billion for any one event.
- 2.2 Cover for charterers' co-assured under an owner's entry shall be subject to a limit for any one event of USD 350 million in the aggregate for all claims arising in respect of the actual or threatened escape or discharge of oil.
- 2.3 Where the member and/or any other party or parties interested in the operation of the vessel are insured under separate owners' entries with the Association or any other insurer which participates in the Pooling Agreement and/or the International Group Re- Insurance policies, the aggregate of all claims brought against the Association and/or other insurers shall be limited to the sum set out in paragraph

- 2.1. The liability of the Association in respect of each such claim shall be limited to that proportion of the sum referred to in paragraph 2.1. that the claim recoverable from the Association bears to the aggregate of the claims recoverable from the Association and the other insurers.
- 2.4 Where the entered vessel provides salvage or other assistance to another vessel following a casualty, a claim by the member in respect of oil pollution arising out of the salvage or other assistance or the casualty shall be aggregated with any claims in respect of oil pollution by any other vessel similarly engaged in connection with the same casualty when such other vessels are insured for pollution risks under owner's entries in the Association or any other insurer which participates in the Pooling Agreement and/or the International Group Re-Insurance policies. In such circumstances, the Association's liability shall be limited to such proportion of the sum referred to in paragraph 2.1 as the claim by the member bears to the aggregate of all the said claims

3. Limitation - charterers' entries

Claims under charterers' entries in respect of claims arising out of the actual or threatened escape or discharge of oil shall be included in and subject to the limitations contained in rule 32.3.

4. Other insurance

Where the member has insurance for liability, loss, costs and fines arising out of oil pollution which would otherwise fall within the cover provided by these Rules, such cover shall be excluded up to the stated limit of such insurance and the sum insured under such other insurance shall be deducted from the amount to which the Association is entitled to limit its liability.

APPENDIX 5A PASSENGERS, CREW AND OTHER PERSONS CARRIED ON BOARD

- For the purposes of this Appendix, and without prejudice to anything else contained in these Rules, a "Passenger" shall mean a person carried on board a vessel under a contract of carriage or who, with the consent of the carrier, is accompanying a vehicle or live animals covered by a contract for the carriage of goods and "Crew" shall mean any other person on board a ship who is not a Passenger.
- 2. Unless otherwise limited to a lesser sum, the Association's aggregate liability arising under any one owner's entry shall not exceed
 - 1) in respect of liability to Passengers USD 2 billion any one event; and
 - (2) in respect of liability to Passengers and Crew USD 3 billion any one event.
- Provided always that where there is more than one owner's entry in respect of the same ship in the Association and/or by any other insurer which participates in the Pooling Agreement
 - (1) the aggregate of claims in respect of liability to Passengers recoverable from the Association and/or such other insurers shall not exceed USD 2 billion any one event and the liability of the Association shall be limited to such proportion of that sum as the claim recoverable by such persons from the Association bears to the aggregate of all such claims otherwise recoverable from the Association and all such insurers; and
 - (2) the aggregate of all claims in respect of liability to Passengers and Crew recoverable from the Association and/or such other insurers shall not exceed USD 3 billion any one event and the liability of the Association shall be limited
 - (i) where claims in respect of liability to Passengers have been limited to USD 2 billion in accordance with proviso (1) to such proportion of the balance of USD 1 billion as the claims recoverable by such persons in respect of liability to Crew bears to the aggregate of all such claims otherwise recoverable from the Association and all such insurers; and
 - (ii) in all other cases, to such proportion of USD 3 billion as the claims recoverable by such persons in respect of liability to Passengers and Crew bears to the aggregate of all such claims otherwise recoverable from the Association and all such insurers.



APPENDIX 6 CONTRACTS AND CONTRACTUAL TERMS

1. Terms required by the Association in contracts

of carriage of cargo

- 1.1 The Association requires the member to incorporate the Hague Rules or the Hague-Visby Rules or corresponding provisions in all contracts of carriage where permitted by relevant law.
- 1.2 Association requires the member to incorporate the "New Jason Clause" and "Both to Blame Collision Clause" in all contracts of carriage used in international trade.

2. Other terms

The Association may during the policy year require the member (for the purposes of Rule 29.1.8) to incorporate into any contract or indemnity any term which the Association in its discretion considers should be so incorporated.

3. Towage contracts approved by the Association

3.1 Towage of the entered vessel (Rule 23.1.2)

It is a requirement of approval under Rule 23.1.2 that the contract incorporates a term to the effect that the member and the owner of the towing vessel shall each be responsible for any loss or damage to his own vessel and for loss of life or personal injury on his own vessel without any recourse whatsoever against the other.

3.2 Towage by the entered vessel (Rule 23.4)

Where the vessel is engaged in towage under Rule 23.4 under an owners' entry,, the Association approves the following contracts:-

- a) Scandinavian, UK or Netherlands standard towage conditions.
- b) "Towcon" or "Towhire".
- c) Lloyd's Standard Form of Salvage Agreements, and
- d) other contracts containing similar exclusions of liability clauses to those in the contracts specified in (a), (b) and (c) or incorporating a term between (i) the member and (ii) the owner of the tow and the owners of any cargo or other property on board the tow, that each shall be responsible for loss of or damage to, or injury or death of, its own property and personnel and property and personnel of its affiliates and other contractors (and in the case of off-shore operators, co-licensees) and its and their sub-contractors, irrespective of any fault or neglect of that party or its affiliates, contractors or co-licensees or its or their sub-contractors or of any of their personnel (i.e. a knock-for knock).

4. Heavy lift contracts approved by the Association

The "Heavycon" charterparty is approved by the Association under an owners' entry provided it is not materially amended.

APPENDIX 7 ADDITIONAL INSURANCES

If separately agreed, the Association can arrange insurance for additional risks. The following is a brief description of some of the covers available. Full details of these additional insurances (including applicable limits) can be obtained from the Association on request

These insurances shall be subject to the provisions contained in the Rules, unless otherwise agreed.

Charterers' liability for loss of or damage to the entered vessel

The Association can arrange insurance for charterers' liability to the entered vessel, including for loss of or damage to the entered vessel, her equipment, stores and supplies, demurrage or loss of use of the entered vessel and contributions to general average and salvage.

2. Combined charterers' cover

The Association can arrange combined insurance for charterers to cover P&I insurance in accordance with Part II of the Rules and liability for loss of or damage to the entered vessel.

3. Charterers' bunkers

The Association can arrange for insurance of charterers' bunkers carried on board the entered vessel.

4. Deviation

The Association can arrange additional insurance to cover members' liability arising out of:-

- 4.1 geographical deviation or other departure from the contractually agreed voyage (such as transhipment, dry-docking with cargo on board, or lightering) which deprives the member of the right to rely on defences or rights of limitation which would otherwise be available, and
- 4.2 carriage of cargo on deck on under deck bills of lading or similar documents which should have been, but were not, claused to show that the cargo was carried on deck. The insurance is subject to the following conditions,
 - a) the member must notify the Association before the deviation takes place or as soon as he becomes (or should become) aware of it,
 - b) the Association must approve the deviation, and
 - c) the member must pay any additional premium required.

5. Storage of cargo

The Association can arrange insurance to cover members' liability for cargo loss, damage or shortage occurring more than 21 days before loading or after discharge while the cargo is stored on a wharf or in a warehouse or lighter. The member must notify the Association as soon as he becomes (or should become) aware that cargo is likely to be stored in the circumstances described above, and pay any additional premium required.



6. Container risks

The Association can arrange insurance for

- 6.1 loss of or damage to containers owned or leased by the member, and
- 6.2 the member's liability to third parties arising from an accident during use of the container.

7. Through transport

The Association can arrange insurance to cover the member's liability for loss, shortage, damage or delay to cargo whilst in the custody of another carrier. Cover is limited to liabilities which arise under a contract which provides for carriage to be performed partly by the entered vessel, and has been approved by the Association.

8. Crew familiarisation and superintendents

The Association can arrange insurance to cover the member's liability to third parties (excluding employers' liability) arising out of the presence on board of officers or crew prior to or after delivery for purposes of familiarisation, or as superintendents on a new building which is to be entered in the Association.

9. Salvage

The Association can arrange insurance for an owner member who is a professional salvor for liability costs and expense arising out of salvage operations

- 9.1 in connection with the operation of and in respect of the member's interest in the entered vessel, or
- 9.2 in the absence of an entered vessel, in respect of
 - a) oil pollution risks, and
 - b) risks other than oil pollution.

10. Supply and diving support vessels

The Association can arrange insurance to cover liabilities for supply, diving support and similar vessels.

11. Freight

The Association can arrange insurance for loss of freight.

12. Kidnap & ransom

Kidnap and ransom insurance - including a loss of hire extension.

13. War Risks

The Association has arranged additional insurance to cover war risks liabilities which are excluded by Rule 30.4.1, and are in excess of the USD equivalent of the proper value of the entered vessel or the amount recoverable from the vessel's War Risks Underwriters whichever is the greater. The cover is subject to a minimum deductible of USD 50.000. For the purpose of this cover, where the proper value exceeds USD 100 million, the proper value shall be deemed to be USD 100 million.

The insurance excludes claims recoverable under

- a) Hull insurance.
- b) Marine war risks and any P&I inclusion clauses attached, and
- c) Norwegian or similar state or government schemes or specific policies covering liabilities to crew

The insurance may be terminated on 7 days' notice or, in some circumstances automatically.

14. Chemical, Bio-Chemical Electromagnetical Weapons and Computer Virus Risks The Association has arranged cover for

- a) damages, compensation or expenses in consequence of the personal injury to or illness or death of crew, and
- b) legal costs and expenses incurred solely for the purpose of avoiding or minimising any other P&I liability (other than under the Omnibus Rule) arising out of chemical, biological, bio-chemical, electromagnetical weapons and computer virus risks (as defined in Appendix 1).

Cover is subject to aggregation for all interests in the vessel and is subject to limits, cancellation and geographical exclusions.

NORWEGIAN STATUTES

FORENINGENS VEDTEKTER

The Norwegian version of the Statutes which appear in the English language at the front of this book.

1. FORENINGENS FORMÅL OG ORGANISASJON

1.1 Stiftelse og formål

- 1.1.1 Assuranceforeningen SKULD (Gjensidig) er stiftet i januar 1897, og har sitt hovedkontor i Oslo, med kontorer i Bergen, København, Hamburg, Hong Kong og Pireus.
- 1.1.2 Som medlemmer kan opptas rederier, eiere, operatører, disponerende redere, forsikringsgivere og befraktere av skip, derunder slike flytende innretninger som styret til enhver tid måtte bestemme.
- 1.1.3 Foreningens formål er gjensidig forsikring av ansvar og tap som medlemmene påtar seg i direkte forbindelse med driften av de inntegnede skip derunder risiko som knytter seg til slik forsikring såsom rettsvernsforsikring, befrakterforsikring og forsikring av flytende flyttbare enheter og lignende innretninger.
- 1.1.4 Foreningen kan delta i reassuranse- og koassuranse arrangementer i samsvar med Foreningens formål.
- 1.1.5 Foreningen kan opprette datterselskaper for å utøve virksomhet i samsvar med Foreningens formål.
- 1.1.6 Foreningen kan anta et driftsselskap.
- 1.1.7 Medlemmene er ikke ansvarlige overfor andre medlemmer eller overfor tredjemann for Foreningens forpliktelser.
- 1.1.8 Ved uttreden av Foreningen er medlemmet ikke berettiget til noen andel av Foreningens egenkapital eller overskudd.
- 1.1.9 Foreningens forsikrings- og regnskapsår løper til kl. 12.00 GMT den 20.2. hvert år.
- 1.1.10 Medlemmene er sammen med Foreningen også medlemmer av Skuld Mutual Protection and Indemnity Association (Bermuda) Ltd. Medlemskapet er undergitt bestemmelsene i Bermuda lov 1978 nr. 16 med senere endringer.

1.2 Generalforsamlingen

- 1.2.1 Generalforsamlingen er Foreningens høyeste myndighet.
- 1.2.2 Ordinær generalforsamling avholdes senest innen 6 måneder etter regnskapsårets utløp. Ekstraordinær generalforsamling holdes når styret eller representantskapet finner det nødvendig, eller når minst 10 medlemmer med tilsammen minst 20 stemmer forlanger det.
- 1.2.3 Generalforsamling kan i tillegg til i Oslo holdes i København, Pireus, Hong Kong, Singapore, London og Hamilton, Bermuda.
- 1.2.4 Generalforsamlingen innkalles av styret med minst 14 dagers skriftlig varsel. Med innkallelsen skal følge representantskapets uttalelse vedrørende de spørsmål som skal tas opp til behandling og de øvrige dokumenter som styret finner nødvendig. Innkallelsen til den ordinære generalforsamling vedlegges styrets forslag til årsoppgjør med revisors beretning og representantskapets og kontrollkomiteens uttalelse.
- 1.2.5 Generalforsamlingen kan bare treffe beslutning i saker som er angitt i innkallelsen og som representantskapet har hatt anledning til å uttale seg om.
- 1.2.6 Generalforsamlingen åpnes av styrets formann, eller i hans fravær av en person oppnevnt av styret. Generalforsamlingen velger møteleder. Generalforsamlingen er beslutningsdyktig uten hensyn til hvor mange medlemmer som er representert.

1 til og med	20.000 én stemme
til og med	50.000 totalt to stemmer
til og med	100.000 totalt tre stemmer
til og med	200.000 totalt fire stemmer
til og med	400.000 totalt fem stemmer
til og med	600.000 totalt seks stemmer
til og med	800.000 totalt syv stemmer
til og med	1.000.000 totalt åtte stemmer

Tonnasje over 1.000.000 tonn gir ytterligere én stemme pr. 500.000 tonn.

- 1.2.8 Inntegning for kortere tid enn 1 år eller inntegning som ikke er undergitt regnskapsmessig tilskuddsplikt, gir ikke stemmerett. Stemmeretten beregnes på basis av inntegnet tonnasje pr. den dag innkallelsen sendes medlemmene. Stemmeretten faller dog bort for skip hvor inntegningen senere er opphørt.
- 1.2.9 Medlemmer hvis skip bestyres av samme disponentfirma eller forsikres av samme forsikringsgiver, har tilsammen så mange stemmer som de ville ha hatt om samtlige innmeldte skip hadde tilhørt ett medlem.
- 1.2.10 Ethvert medlem kan la seg representere ved fullmektig i generalforsamlingen. Som fullmektig kan kun velges medlemmer med stemmerett etter 1.2.8. Ingen kan avgi stemme på egne eller andres vegne med tilsammen mer enn 1/10 av de tilstedeværende stemmer.
- 1.2.11 I forbindelse med at det innkalles til generalforsamling, kan det kreves at møtedeltakerne forhåndsanmelder på hvis vegne de møter. Skriftlig melding om dette må være innkommet innen kontortidens utløp to dager forut for generalforsamlingen.
- 1.2.12 Beslutninger treffes med simpelt flertall. Dog kreves 2/3 flertall til vedtagelse av lovendringer. Ved stemmelikhet gjør møtelederens stemme utslaget.

1.3 Ordinær generalforsamlings gjøremål

Den ordinære generalforsamling skal:

- 1.3.1 Fastsette resultatregnskap og balanse, godkjenne styrets beretning og treffe beslutning om innkalling av tilskudd, refusjon av for meget innbetalt forskuddspremie eller utdeling av overskudd. Generalforsamlingen kan ikke beslutte innkalt lavere tilskudd, refusjon av større andel av forskuddspremie eller utbetalt større overskuddsandel enn anbefalt av representantskapet.
- 1.3.2 Behandle kontrollkomiteens rapport.
- 1.3.3 Fastsette antall medlemmer i representantskapet.
- 1.3.4 Velge representantskapets medlemmer og fastsette deres godtgjørelse.
- 1.3.5 Velge medlemmer til valgkomitéen.
- 1.3.6 Velge Kontrollkomitéens medlemmer og fastsette deres godtgjørelse.
- 1.3.7 Treffe beslutning om avslutning av forsikringsår.
- 1.3.8 Treffe beslutning om endringer av Foreningens Lover.
- 1.3.9 Treffe beslutning om andre saker som hører under generalforsamlingen og som er angitt i innkallelsen.

1.4 Representantskapet

- 1.4.1 Representantskapet skal bestå av ordfører og varaordfører samt av minst 10 og høyst 28 medlemmer som fastsettes av generalforsamlingen. Representantskapet velges av generalforsamlingen etter innhenting av innstilling fra valgkomitéen. Representantskapets ordfører og varaordfører velges ved særskilt valg.
- 1.4.2 Representantskapet er beslutningsdyktig når minst halvparten av medlemmene, derunder ordfører eller varaordfører, er tilstede.
- 1.4.3 Representantskapets møter holdes så ofte representantskapets ordfører finner det nødvendig eller når minst 1/3 av medlemmene eller styret krever det.
- 1.4.4 Foreningens styre og administrerende direktør deltar på representantskapets møter medmindre annet bestemmes for en bestemt sak. Deltagelsen skjer uten stemmerett.

1.5 Representantskapets gjøremål

- 1.5.1 Representantskapet skal føre tilsyn med styrets og administrerende direktørs forvaltning av Foreningen og påse at dens formål fremmes i samsvar med lovgivning og Foreningens lover samt generalforsamlingens og representantskapets egne beslutninger. Hvert av medlemmene kan på møter i representantskapet kreve opplysninger om Foreningens drift. Representantskapet kan selv, eventuelt ved oppnevnte personer, iverksette undersøkelser.
- 1.5.2 Representantskapet velger etter anbefaling fra valgkomiteen styret og styrets formann og viseformann og fastsetter deres godtgjørelse. Formann og viseformann velges separat.
 Representantskapet skal også etter anbefaling fra valgkomiteen velge styre for alle Foreningens datterselskap og tilknyttede selskaper.
- 1.5.3 Representantskapet skal videre:
 - a) Gi uttalelse til generalforsamlingen om godkjennelse av styrets forslag til årsregnskap og styrets beretning samt styrets forslag til anvendelse av overskudd eller dekning av tap herunder utligning av tilskudd, refusjon av for meget innkalt forskuddspremie eller utdeling av overskudd.
 - Fremlegge for generalforsamlingen sin innstilling angående avslutning av forsikringsår.
 - c) Fremlegge for generalforsamlingen sin innstilling angående lovforslag anbefalt av styret.
 - d) Fastsette instruks for kontrollkomitéen og behandle dens innberetning.
 - e) Velge revisor og godkjenne hans godtgjørelse.
 - f) Godkjenne Foreningens driftsavtaler.
 - g) Behandle andre saker som hører inn under representantskapet og som er inntatt i sakslisten.

1.6 Kontrollkomitéen

- 1.6.1 Kontrollkomitéen skal bestå av minst 3 medlemmer og ett varamedlem valgt av generalforsamlingen.
- 1.6.2 Kontrollkomitéens formann velges hvert år av og blant kontrollkomitéens medlemmer.
- 1.6.3 Kontrollkomitéen kan avholde møte når minst tre av medlemmene er tilstede, og innkalling er sendt med rimelig frist.

1.7 Kontrollkomitéens gjøremål

Kontrollkomitéen skal i henhold til instruks vedtatt av representantskapet føre tilsyn med Foreningens virksomhet og påse at den følger lover, forskrifter og vilkår samt vedtak i Foreningens besluttende organer. Kontrollkomitéen skal minst en gang i året gi innberetning til representantskapet og generalforsamlingen.

1.8 Styret

- 1.8.1 Styret skal påse at Foreningens og medlemmenes interesser blir ivaretatt på forsvarlig måte og treffe de beslutninger som ikke er tillagt representantskap eller generalforsamling.
- 1.8.2 Styret består i henhold til beslutning av representantskapet av 5 til 9 medlemmer innbefattende en formann og viseformann. Medlemmene velges uten varamenn i samsvar med Lovenes 1.5.2. I tillegg skal to medlemmer valgt av og blant ansatte av Foreningen eller et datterselskap oppnevnt som driftsselskap sitte i Styret.
- 1.8.3 Styret møter etter innkalling fra styrets formann. Han kan bestemme at saker behandles skriftlig eller på annen tilfredsstillende måte. Medlem av styret og administrerende direktør kan kreve at styret behandler en særskilt sak.
- 1.8.4 Styret er beslutningsdyktig når flere enn halvparten av styrets medlemmer er tilstede eller deltar i behandlingen av en sak. For at et vedtak skal være gyldig, må de som stemmer for beslutningen utgjøre mer enn 1/3 av samtlige medlemmer. Ved stemmelikhet gjør formannens, eller i hans fravær viseformannens, stemme utslaget.
- 1.8.5 Foreningens administrerende direktør (CEO) deltar i styrets møter, medmindre styret for den enkelte sak treffer annen beslutning. Han møter uten stemmerett.

1.9 Styrets gjøremål

Styret skal:

- 1.9.1 Lede Foreningens virksomhet.
- 1.9.2 Ansette Foreningens administrerende direktør (CEO) og fastsette hans instruks og vilkår.
- 1.9.3 Fatte beslutninger vedrørende avdelingskontorer og datterselskaper og fastsette de generelle retningslinjer for deres aktivitet. Hvor et avdelingskontor er etablert med lokalt styre, skal retningslinjene fastsettes i samråd med og etter innstilling fra dette. Det lokale styres mandat skal følge de retningslinjer som er gitt av Foreningens styre.
- 1.9.4 Fremlegge for representantskapet fullstendig og revidert årsregnskap og årsberetning, herunder innstilling om avslutning av forsikringsår, utligning av tilskudd, refusjon av for meget innkalt forskuddspremie eller utdeling av overskudd.
- 1.9.5 Fremlegge for representantskapet anbefaling angående lovforslag.
- 1.9.6 Fatte beslutning om endringer i Foreningens forsikringsvilkår.
- 1.9.7 Godkjenne Foreningens reassuransekontrakter.
- 1.9.8 Forvalte Foreningens midler overensstemmende med vedtatte retningslinjer.
- 1.9.9 Fatte beslutning om opphørstilskudd.
- 1.9.10 Fastsette de generelle premier som skal gjelde i forsikringsåret.
- 1.9.11 Fastsette retningslinjer for godkjennelse av erstatningsoppgjør.

- 1.9.12 I det enkelte tilfelle treffe beslutning om at Foreningen skal erstatte ansvar eller tap utover det som følger av Foreningens forsikringsvilkår eller som er unntatt fra dekning når styret ut fra Foreningens formål finner dette naturlig og ønskelig.
- 1.9.13 Fremlegge for representantskapet og generalforsamlingen anbefalinger i enhver sak som er hatt til behandling.

1.10 Foreningens tillitsmenn

- 1.10.1 Valgbar som medlem av representantskapet og styret er enhver som etter et medlems ønske skal representere medlemmet som tillitsmann i Foreningen. Hertil kan inntil én person, som ikke forøvrig er valgbar, velges som medlem av styret.
- 1.10.2 Valgbar som medlemmer av kontrollkomitéen er enhver som etter et medlems ønske skal representere medlemmet med unntak av representantskapets ordfører og varaordfører samt medlemmer av styret. Hertil kan inntil en person som forøvrig ikke er valgbar innvelges. Et av medlemmene må tilfredsstille domstollovens krav til å være dommer. Valg av dette medlem skal godkjennes av Kredittilsynet.
- 1.10.3 I representantskap, kontrollkomité og styre er medlemmenes funksjonstid normalt 2 år, således at tilnærmet halvparten av medlemmene trer ut hvert år. Gjenvalg kan skje.
- 1.10.4 Medlemmer av styre og representantskap som ikke lenger er valgbare i henhold til 1.10.1 kan ikke gjøre tjeneste.
- 1.10.5 Ingen kan samtidig gjøre tjeneste i både styret og representantskapet. Ingen kan bli valgt eller gjenvalgt etter fylte 67 år. Vedkommende kan imidlertid gjøre tjeneste i den gjenværende valgperiode.

1.11 Valgkomitéen

- 1.11.1 Foreningens valgkomité skal bestå av representantskapets ordfører samt 2 medlemmer valgt av generalforsamlingen. Administrerende direktør (CEO) fungerer som sekretær med møte- og talerett. Valgkomiteen velger selv sin formann.
- 1.11.2 Valgkomitéen avgir innstilling om valg av samtlige tillitsmenn til Foreningen.
- 1.11.3 I sin innstilling skal Valgkomiteen ta hensyn til at sammensetningen av styre og representantskap skal avspeile Foreningens internasjonale aktivitet, dens innmeldte tonnasje og kundestruktur og forøvrig være allsidig sammensatt.
- 1.11.4 Styret skal godkjenne retningslinjer for Valgkomitéen.

2. TVISTER

2.1 Voldgift og lovvalg

- 2.1.1 Disse lover (vedtekter) er underlagt norsk rett dog slik at forsikrings-avtaleloven av 1989 ikke får anvendelse.
- 2.1.2 Hvis ikke annet er avtalt, skal tvister mellom Foreningen og et medlem avgjøres ved voldgift. Det samme gjelder behandlingen av tvister mellom Foreningen og et tidligere medlem, og mellom Foreningen og en som er eller har vært forsikret eller medforsikret under Foreningens polise uten å være eller ha vært medlem.

- 2.1.3 Hver av partene oppnevner en voldgiftsmann, og disse oppnevner i fellesskap voldgiftsrettens formann. Hvis de ikke blir enige om oppnevnelsen, eller en part unnlater å foreta oppnevnelse, skal Justitiarius ved Oslo Tingrett foreta oppnevnelsen.
- 2.1.4 Forhandlingene ved voldgiftsretten skal finne sted i Oslo og norsk rett skal legges til grunn. Voldgiftsavgjørelser skal grunngis.

3. OPPLØSNING OG OVERGANGSREGLER

3.1 Oppløsning av Foreningen

Oppløsning av Foreningen eller sammenslutning med andre foreninger kan besluttes av generalforsamlingen med 2/3 flertall av samtlige medlemmers stemmer, beregnet som nevnt i 1.2.7 flg. Er mindre enn 2/3 av stemmene representert, kan saken forelegges en ny generalforsamling, og oppløsning eller sammenslutning kan vedtas med 3/4 flertall av de representerte stemmer. Mulig overskudd som fremkommer etter at Foreningens forpliktelser er avviklet skal, hvis Foreningen oppløses, disponeres av generalforsamlingen. Ved sammenslutning kan overskuddet etter generalforsamlingens beslutning disponeres som nevnt eller overføres til den nye forening.

3.2 Overgangsregler

Disse lover trer i kraft og de tidligere lover oppheves fra 20.2.2011 kl. 12.00 GMT. Ansvar eller tap som er oppstått før det nevnte tidspunkt, bedømmes etter de tidligere lover.

INDEX TO RULES



INDEX TO RULES

Subject: Rule No. Ad valorem bill of lading 5.2.10, 5.4.4 **Affiliate** 1.2, 32.4, 45, Appendix 1, Appendix 7 Arbitration 47 30.4.1, 42 **Assignment** Biochem 32.5, Appendix 1, Appendix 7 Bolero See Paperless trading 30.4.1, Appendix 1 **Bunkers Convention Certificate** Burden of proof 31 4 Calls 4.2 - annual - supplementary 4.3 4.4, 32.2, Appendix 3 - overspill - release 4.5 Cargo 5 - deviation 5.2.5, 5.2.11, 5.4.5, Appendix 7 - storage Appendix 7 5.2.5, 5.4.5, Appendix 7 - deck cargo - lightering of cargo 5.2.3 - through transport 5.2.3, 5.2.8, 5.4.1, Appendix 7 - fines 19.1.2-3 - extra handling costs 4 Cash. valuables 30.1.6 5.2.9. 5.4.3 - cargo - crew 7.2.3, 7.3.2 8.2.2, 8.3 - passengers Certificate of Entry 1, Appendix 2 Cessation of cover Charterer 1, Appendix 1, Appendix 7 12.5, Appendix 7 - liability to hull - limitation 32 Charterparty 27 - cancellation 30.2.7 Chemical Appendix 1 - biochemical Appendix 1 Claims handling 36 Class action 27.2.4 Classification 3.3.2, 28.4, 29, 35.6 **CLC** Certificate 30.4.1, Appendix 1 Co-assured 1, 32.4, 45, Appendix 1, Appendix 2 12 Collision Conditions 2 - conditions precedent 28 - other conditions 29

20

Confiscation

6.1.	B N
Subject:	Rule No.
Contact liability	12
Containers	30.2.4, Appendix 7
Contraband Contracts & Indemnities	30.4.3
	29.1.8-9, 30.2.1, Appendix 6
- cargo	5.2.7 7.2.1
- crew	8.2.1
- passengers Cover	2
- cessation	3.1
- termination	3.2-3
Crew	7, Appendix 1
- fines	19.1.4, 19.4
- limits	7.3, 32.1.7, Appendix 5A
Deductibles	4.12, Appendix 4
Default judgment or award	30.1.9, 28.5
Defence cover	27
Definitions	Appendix 1
Delay	Аррения і
- Cargo	5.1, 5.2.12
- Passengers	8.1.7
- Vessel	30.2.6, 30.2.9
Demurrage	27, 30.2.6
Deviation	5.2.5, 5.2.11, 5.4.5, Appendix 7
Disclosure	28.1
Diversion costs	11
Documentation & information	29.1.5, 36.2.2-3
Drilling operations	30.3.1, Appendix 1
Duty	4.11.4
Electronic Trading	5.2.17, 30.4.5, Appendix 1
Enquiries	25.2
Entered vessel	3, Appendix 1
- loss of or damage to	12.5, 30.2.3-5
- charterer's cover for liability	to 12.5, Appendix 7
- wreck removal	15
Estimated Total Calls (ETC)	4.2
Excluded operations	30.3
Exclusions	30
- general exclusions	30.1
- excluded losses	30.2
- excluded operations	30.3
- excluded risks	30.4
- of liability	40
Extra cargo handling costs	6
FF0	12
Fines	19,20
Flag state requirements	29.1.4

- limits

Persons carried on board

Persons saved at sea

Persons not carried on board

Subject: Rule No. Fleet entry 1.2, 45, Appendix 1 FMC Undertaking 30.4.1, Appendix 1 Fraudulent acts 3.3.2, 28.2 Freight 27, 30.2.6, Appendix 7 General Average 17,18 Hague, Hague/Visby Rules 5.2.6, 5.2.12, Appendix 1, Appendix 6 Hazardous Trading 30.4.3 Heavy lift 5.2.18, 15.2.2, Appendix 6 Hire 27, 30.2.6 12, 30.1.2, Appendix 1 Hull insurance Improper trading 30.4.3 30.4.3 Imprudent trading Indemnities, see Contracts Insolvency 3.1, 30.1.5 Interest 4.11.4, 44 ISM Code 28.4, 29.1.4, 35.6, Appendix 1 28.4, 29.1.4, Appendix 1 ISPS Code Landfills 30.4 4.7 Lay-up returns Legal costs 25, 27, 39.5 Lien 4.13 Limitation 2,32 14, 32.1.6, Appendix 5 - oil pollution - defence 27.4 Member 1 - Joint member 1, 32.4, 45 Member's own property 26 Misconduct 30.1.8, Appendix 1 Mitigation 24,34 Mortgage 42 **Nuclear Risks** 30.4, Appendix 1 Obstruction 16 **Omnibus** 46 Other insurance 30.1.3-4 Appendix 5 - pollution Overspill 4.4, 32.2, Appendix 3 Owner 1, Appendix 1 - limitation 32 Paperless trading See Electronic trading **Passengers** 8. Appendix 1 - limits 8.3, 32.1.7, Appendix 5A Pay to be paid 28.5

9.3, 32.1.7, Appendix 5A

11.1.2. 21.1.1-2

10

Subject:	Rule No.
Policy Year	1, 4.10, Appendix 1
Pollution	14, Appendix 5
- U.S. oil pollution	14, 30.4.4, Appendix 5
- fines	19, Appendix 5
- limitation	14.3, 32.1.6, Appendix 5
Premiums	3.3.2, 4, 28.3
- see also Calls	
- fixed premiums	4.6
Production operations	30.3.1, Appendix 1
Property liability	13
- Member's own property	26
Quarantine	21
Refugees	11, 30.2.8
Reserves	4.9
Risks	2.2
- disclosure	28.1.1-2
- alteration	28.1.3
- excluded	30.4
- burden of proof	31
Safety regulations	29.1.2, 29.1.4
Sanctions	30.4.7, 32.7
Salvage	22, 30.2.2, Appendix 7
Savings of the member	41
SCOPIC	22.1.3
Security	38, 39.2
Set-off	43
Special Compensation	22
STOPIA	14, 30.4.1, Appendix 1
Specialist operations	30.3.3, Appendix 1
Stowaways Sub-sea activities	11 20.2.2. Appendix 1
Sub-sea activities Sue and labour	30.3.2, Appendix 1 24
	4.8
Surplus Surveys	3.3.2, 29.1.1, 29.1.6, 30.1.1, 35
Taxes	4.11.4
Termination of cover	3
Terrorism - see War Risks	3
Through transport	5.2.3, 5.2.8, 5.4.1, Appendix 7
Time limits	37
TOPIA	14.1.4, 14.5, Appendix 1
Towage	23, 30.2.2, Appendix 6
Unlawful acts	3.3.2, 28.2
Unlawful trading	30.4.3
Unsafe trading	30.4.3
Unseaworthiness of entered	50.4.5
vessel	3.3.2, 30.1.7, 35.4
vC33El	5.5.2, 50.1.7, 55.4



Subject:

U.S. oil pollution Valuables, see Cash

Vessel

- see Entered Vessel

Waiver War risks

Waste disposal activities

Wilful Misconduct

Wreck removal

Rule No.

14, 30.4.4, Appendix 5

Appendix 1

39

30.4.1, 31, 32.5, Appendix 1, Appendix 7

30.3.2, Appendix 1

30.1.8, Appendix 1

12.1.2,15



